In the Supreme Court of the United States

NEAL BISSONNETTE and TYLER WOJNAROWSKI, on behalf of themselves and all others similarly situated, *Petitioners*, V.

LEPAGE BAKERIES PARK ST., LLC, C.K. SALES CO., LLC, and FLOWERS FOODS, INC., *Respondents*.

On Writ of Certiorari to the United States Court of Appeals for the Second Circuit

JOINT APPENDIX

JENNIFER D. BENNETT GUPTA WESSLER LLP 505 Montgomery Street Suite 625 San Francisco, CA 94111 (415) 573-0336 jennifer@guptawessler.com

250 Vesey Street New York, NY 10281 (212) 326-7830 tlovitt@jonesday.com

TRACI L. LOVITT

JONES DAY

Counsel of Record for Petitioners Counsel of Record for Respondents

PETITION FOR A WRIT OF CERTIORARI FILED: JULY 17, 2023 CERTIORARI GRANTED: SEPTEMBER 29, 2023

TABLE OF CONTENTS

Exhibit 1 to Memorandum in Support of Defendants' Motion to Dismiss or, in the Alternative, to Compel Arbitration: Declaration of Jake Linthicum (Sept. 18, 2019)1
The following opinions, decisions, judgments, and orders have been omitted in printing this joint appendix because they appear on the following pages in the appendix to the Petition for Certiorari:
Initial Opinion of the United States Court of
Appeals for the Second Circuit (May 5, 2022)Pet. App. 1a
Gircuit (114) 9, 2022)
Amended Opinion of the United States Court
of Appeals for the Second Circuit (September 26, 2022)Pet. App. 38a
Order of the United States Court of Appeals for the Second Circuit Denying Rehearing
En Banc (February 15, 2023)Pet. App. 77a
Order of the United States District Court for the District of Connecticut Denying Motion to Dismiss (May 14, 2020)Pet. App. 99a
Judgment of the United States District Court for the District of Connecticut Denying Motion to Dismiss (May 15, 2020)Pet. App. 120a

IN THE UNITED STATES DISTRICT COURT DISTRICT OF CONNECTICUT

)	
)	
)	
)	
)	
)	
)	
)	Civil Action No.
)	3:19-cv-00965-
)	KAD
)	
)	
)	
)	
)	
)	

DECLARATION OF JAKE LINTHICUM

I, Jake Linthicum, hereby depose and state as follows:

1. I am currently employed as the Distributor Enablement Operations Coordinator ("DEOC"), a position that was formerly referred to as the Director of Distributor Relations ("DDR"), for Lepage Bakeries Park Street, LLC ("Lepage") with responsibility for the New England Region, which includes Connecticut. Lepage owns CK Sales Co., LLC ("CK Sales") and is, in turn, a wholly owned subsidiary of Flowers Foods, Inc. I became the DEOC in January 2018 and, before that, worked as the DDR since April 2013.

2. Flowers Foods is the parent holding company of numerous operating subsidiaries, which produce fresh breads, buns, rolls, and snack cakes. Lepage, one such

1

subsidiary, is a baker of fresh breads, buns, and other baked goods products (i.e., a manufacturer of bread and bakery products). CK Sales is in the business of contracting with independent distributors.

3. Lepage uses a direct-store-delivery ("DSD") system whereby independent distributor franchisees, like Plaintiffs Neal Bissonnette and Tyler Wojnarowski, purchase the rights from CK Sales to sell and distribute products to customers in a defined territory.

4. As the DEOC, and previously as the DDR, I am (and was) responsible for, among other things, maintaining distributor files for distributors, including Distributor Agreements and the Arbitration Agreements attached to Distributor Agreements. In addition, I am familiar with various aspects of the DSD distribution system, the distributor business model, distributor product orders, and products produced for distributors.

5. The relationship between CK Sales and distributors, including Plaintiffs Neal Bissonnette and Tyler Wojnarowski, is governed by a Distributor Agreement. Plaintiffs, through their own independent corporations, executed the Distributor Agreement with CK Sales.

6. Neal Bissonnette is a current distributor pursuant to the Distributor Agreement between his corporation, Bissonnette Inc., and CK Sales. Attachment 1 to this Declaration is a true and accurate copy of the Distributor Agreement and Arbitration Agreement signed by Neal Bissonnette, President of Bissonnette Inc., which became effective June 7, 2017.

7. Tyler Wojnarowski is a current distributor pursuant to the Distributor Agreement between his corporation, Blue Star Distributors, Inc., and CK Sales. Attachment 2 to this Declaration is a true and accurate copy of the Distributor Agreement and Arbitration Agreement signed by Tyler Wojnarowski, President of Blue Star Distributors, Inc., which became effective April 25, 2018.

8. Distributors contracted with CK Sales, including Plaintiffs Neal Bissonnette and Tyler Wojnarowski, are responsible for operating their businesses, including hiring employees at their discretion to run their businesses; identifying and engaging potential new customers; developing relationships with key customer contacts; ordering products based on customer needs; servicing the customers in their territory; stocking and replenishing product at the customer locations; removing stale product; and other activity necessary to promote sales, customer service, and otherwise operate their businesses. Because the distributors own the territory, they can build equity in their distributorship businesses. These businesses can appreciate in value and can be sold for a profit. Moreover, under the Distributor Agreements, including those executed by Plaintiffs Neal Bissonnette and Tyler Wojnarowski, distributors do not have to perform any services personally and are contractually obligated to use their "Best Efforts" to increase sales in their territories, which they can do in several ways, including by asking for displays, providing good customer service, recommending new products, soliciting new accounts, and effective merchandising, among other things.

9. Distributors, including Plaintiffs Neal Bissonnette and Tyler Wojnarowski, purchase products from CK Sales, which they ultimately resell to their customers for a higher price. The difference between the price at which they sell the products less the price at which they purchase them, less their business expenses, represents Plaintiffs' "profit margin." Because Plaintiffs and other distributors are compensated based on their sales of products, it is in their best interests to devote necessary time to increasing their sales in their accounts on a daily basis.

10. Plaintiffs Bissonnette and Wojnarowski do not cross state lines to deliver goods in connection with the operation of their businesses.

11. I am aware that in March 2017, a lawsuit involving Connecticut-based distributors, the *Bokanoski* lawsuit, settled. In connection with the *Bokanoski* settlement, distributors who wished to retain their businesses signed a new Distributor Agreement containing an Arbitration Agreement with a class action waiver, in exchange for consideration offered as part of the settlement terms. Other distributors sold their businesses in connection with the settlement and released any and all claims at that time.

12. All distributors that first contracted with CK Sales after the March 2017 *Bokanoski* settlement through the present signed the same type of Distributor Agreements containing Arbitration Agreements with a class action waiver.

13. Distributors who purchase territories from existing distributors are likewise subject to this Arbitration Agreement given their execution of an assumption agreement when they purchase the distributorship, under which they agree to assume all obligations associated with the newly-purchased distributorship, including the Arbitration Agreement.

14. Based on the foregoing, each potential class member, except for two, in this lawsuit is subject to the Arbitration Agreement as follows:

Name	or	Warehouse	<i>Fartucipated in</i> <i>Bokanoski</i> Settlement and Signed Arbitration Agreement in Connection with Settlement	Farticipated in Bokanoski SettlementAlter DokanoskiBokanoski SettlementSettlement and Signed or are Subject to a New ArbitrationArbitration Agreement in Connection with SettlementAlter DokanoskiDistributorship1Settlement
Name		Warehouse	Settlement	Distributorship
Michael Kivera- Negron	Active	Windsor		X
Eldis Senderovic	Active	Windsor		X^*
Roy Rodriguez	Active	Windsor	X	
Tyler Wojnarowski	Active	Waterbury		X
Joe Bokanoski	Active	Windsor	X	
Neal Bissonnette	Active	Waterbury		X
Kyle Sullivan	Active	Waterbury		X
Danny Burgos-Javier	Active	Waterbury		X
Roshawn Telfer	Active	Bridgeport		X
Bruce Hessleton	Active	Bridgeport		X

 1 Individuals who assumed arbitration agreements via execution of an assumption agreement when they purchased their distributorships are identified with an asterisk.

	Χ	N/A	Inactive	Jared Smith
	X	N/A	Inactive	Wayne Nogler
Х		Putnam	Active	Benjamin Nelson
X		Bridgeport	Active	Miguel Otero
X		Waterbury	Active	Mark Murray
X^*		Putnam	Active	David Stockford
X*		Putnam	Active	Joseph Felion
X^*		Bridgeport	Active	Michael Rodriguez
	X	Bridgeport	Active	Joseph Grogan
	X	Bridgeport	Active	Louis Galli
	X	Bridgeport	Active	Ivan Reyes
X		Bridgeport	Active	Joseph Allen
X^*		Waterbury	Active	Richard Kreidel
*X		Bridgeport	Active	Zeleh Kolubah
X		Bridgeport	Active	Anthony Calderon
X^*		Bridgeport	Active	Ana Villegas
Distributorship ¹	Settlement	Warehouse	Status	Name
Purchase of	5		Distributor	
in Connection with	ב			
Arbitration Agreement				
or are Subject to a New	and Signed			
After Bokanoski	Participated in			
Became a Distributor				

Χ		N/A	Inactive	Brandon Hall
X		N/A	Inactive	Thomas Cunningham
X		N/A	Inactive	Tramain Williams
X		N/A	Inactive	Hector Irizarry
X		N/A	Inactive	John Elias
*X		N/A	Inactive	Christopher Williams
*X		N/A	Inactive	Kurtley Roberts
	X	N/A	Inactive	Kathy Bergeron
	X	N/A	Inactive	Charles Sperow
Distributorship ¹	Settlement	Warehouse	Status	Name
Purchase of	à		Distributor	
in Connection with				
Arbitration Agreement				
or are Subject to a New	and Signed			
Settlement and Signed	ler			
After Bokanoski	Participated in			
Became a Distributor				

 $\mathbf{7}$

15. Joshua Reynolds, a distributor from December 14, 2016 to July 27, 2017 is not subject to an arbitration agreement of which Defendants are aware at this time. Additionally, a current distributor named Amy Longo purchased Mr. Reynolds' business and, in doing so, assumed his obligations via execution of an assumption agreement. Like Mr. Reynolds, Ms. Longo is not subject to an arbitration agreement of which Defendants are aware.

Pursuant to 28 U.S.C. § 1746, I declare under penalty of perjury that the foregoing is true and correct, according to my personal knowledge, and if called as a witness, I could and would testify truthfully thereto.

Dated this <u>17th</u> day of September, 2019.

La guio

Jake Linthicum

ATTACHMENT 1

DISTRIBUTOR AGREEMENT TABLE OF CONTENTS

ARTICLE		PAGE
I.	Warranty	1
II.	Definitions	2
III.	Relationship	3
IV.	Sale of Products	4
V.	Best Efforts/Distributor	4
VI.	Best Efforts/Company	6
VII.	Accounts/Trademarks Exempt	6
	From Agreement	
VIII.	Payment for Products	7
IX.	Equipment and Insurance	8
X.	Proprietary Services	9
XI.	Product Delivery	10
XII.	Product Code	10
XIII.	Advertising and Promotions	11
XIV.	Service Requirements	11
XV.	Transfer or Sale of Rights	12
XVI.	Independent Business	13
XVII.	Termination by Company	14
XVIII.	Dispute Resolution	16
XIX.	Trademarks, Trade Names, and	17
	Proprietary Materials	
XX.	Miscellaneous	19
Exhibit A	Territory	23
Exhibit B	Products	24
Exhibit C	Authorized Products	25
Exhibit D	Bill of Sale	26
Exhibit E	Owner	27
Exhibit F	Personal Guaranty	28
Exhibit G	Settlement Statement	29
	Authorization	

Exhibit H	Distributor Choice of Insurance	30
	Coverages	
Exhibit I	Distributor Choice of Delivery	31
	Location	
Exhibit J	Distributor Choice of Fee For	32
	Service	
Exhibit K	Arbitration Agreement	33
Maryland A	ddendum (if applicable)	36
Indiana Add	lendum (if applicable)	37

DISTRIBUTOR AGREEMENT

This Distributor Agreement ("Agreement") is made effective the <u>19th</u> day of <u>June, 2017</u>, by and between <u>CK</u> <u>SALES CO. LLC</u>, with its office and principal place of business at 11 Adamian Drive, Auburn, Maine 04210 ("COMPANY"), and <u>Bissonnette Inc</u>, a *Connecticut* corporation, with its principal place of business at <u>47</u> <u>Harvest Ln Bristol, CT 06010</u> ("DISTRIBUTOR").

W I T N E S S E T H:

COMPANY has developed or acquired a license or franchise for the use of formulae, recipes, trademarks and trade names through which it manufactures and/or markets bread, rolls, and other fresh-baked products, as defined herein; and

Whereas, COMPANY, through the expenditure of time and effort and through its continuing research and marketing programs, has over the years developed a reputation for excellence in quality, value, and superior service which has created strong consumer recognition, approval, and demand for its products; and

Whereas, COMPANY has determined that the divestiture and division of its market area into distribution territories, and the sale of its products to independent franchise distributors for distribution and ultimate retail sale in said territories, will result in a superior distribution system; and

Whereas, DISTRIBUTOR is an Independent contractor with the resources, expertise and capability to act as a franchise distributor of COMPANY's products in the Territory; and

Whereas, the parties desire to create a right in DISTRIBUTOR, under which DISTRIBUTOR will be authorized to sell certain defined products within a

territory or territories and otherwise operate the distributorship business hereunder.

Now, therefore, in consideration of these premises, and of the covenants and conditions contained in this Agreement, and for other good and valuable consideration given and received, the receipt and the sufficiency of which are hereby specifically agreed to and acknowledged, the parties mutually agree as follows:

I. WARRANTY

1.1 The foregoing preambles are Incorporated herein by reference and shall constitute warranties, representations, agreements and undertakings by the parties.

II. DEFINITIONS

- 2.1 *Outlets*: Shall mean all retail stores (except thrift stores) selling food to the general public and all restaurant, fast food, military and institutional accounts, in each case which allow direct store delivery (DSD) Products and/or Authorized Products, except those exempt from this Agreement as set forth in Article VII, below.
- 2.2 **Products**: Shall mean fresh baked goods, as specifically described in **Exhibit B**, attached hereto and made a part hereof, which are currently produced and/or distributed by COMPANY, provided, however, that the Distribution Rights granted herein shall survive only so long as COMPANY and/or its affiliate(s) has/have the legal right to produce and sell such goods, whether by ownership, license, franchise, or otherwise, and COMPANY produces and/or sells such goods. Products shall not include products other than those

specifically described in **Exhibit B** nor products intended to be sold as frozen or refrigerated.

- 2.3 Authorized Products: Shall mean fresh baked goods, as specifically described in Exhibit C, attached hereto and made a part hereof, provided, however, that DISTRIBUTOR does not acquire any proprietary or ownership rights to such Authorized Products, exclusive or non-exclusive. COMPANY may cease selling such Authorized Products to DISTRIBUTOR upon notice, with or without cause. COMPANY may also add Authorized Products to to **DISTRIBUTOR**. Exhibit C upon notice Authorized Products may include products with logos and/or labels, or without.
- 2.4 **Distribution Rights**: Except as expressly limited by this Agreement, Distribution Rights shall mean the right to sell and distribute Products to Outlets in the Territory, which right has been purchased by DISTRIBUTOR from COMPANY as evidenced by a Bill of Sale executed by the parties, which is attached hereto as **Exhibit D**, and made a part hereof.
- 2.5 *Territory*: Shall mean that geographic area as more specifically described in **Exhibit A** within which DISTRIBUTOR owns the Distribution Rights.
- 2.6 Good Industry Practice: Shall mean the standards that have developed and are generally accepted and followed in the baking Industry, including, but not limited to, maintaining an adequate and fresh supply of Products and Authorized Products in all Outlets in the Territory requesting service, actively soliciting all Outlets in the Territory not being serviced, properly rotating all Products and Authorized Products, promptly removing all stale Products and Authorized Products, maintaining proper service and

delivery to all Outlets in the Territory requesting service in accordance with Outlet's requirements, maintaining all equipment in a sanitary condition and in good safe working order, and operating the distributorship business hereunder in compliance with all applicable federal, state and local laws, rules and regulations, including but not limited to, all motor vehicle, Department of Transportation, food, drug, health, bioterrorism, security, and sanitary laws and regulations. **DISTRIBUTOR specifically** acknowledges and agrees that such standards do not reflect control by COMPANY as to the specific details or manner and means of DISTRIBUTOR's business, but reflect only COMPANY's interests in the results achieved by DISTRIBUTOR, protecting the reputation of the brands, and protecting the business reputation of both **DISTRIBUTOR and COMPANY.**

- **2.7** *Stale*: Shall mean Products and Authorized Products removed from the market on the appropriate color code or designated pick up date.
- **2.8** *Out of Code*: Shall mean Products and Authorized Products left in the market beyond the appropriate color code or designated pick up date.
- 2.9 *Chain*: Shall mean a person or business entity that operates more than one Outlet and/or which makes centralized decisions regarding the purchase of Products and/or Authorized Products for more than one Outlet.
- **2.10** *Owner*: Shall mean the individual holding a majority ownership in the stock of DISTRIBUTOR and listed on **Exhibit E** hereto. Such individual must personally guarantee the conditions and obligations herein by executing the Personal Guaranty, attached

hereto as **Exhibit F** and made a part hereof. As long as DISTRIBUTOR owns the Distribution Rights hereunder, Owner agrees to ensure DISTRIBUTOR is properly established, organized and at all times in good standing with the appropriate state agency(s) and in compliance with appropriate state law(s). Upon request, DISTRIBUTOR shall provide COMPANY documentation showing that its corporate registration is in good standing and in compliance with appropriate state law(s).

III. RELATIONSHIP

3.1 *Extent and Duration*: COMPANY hereby recognizes DISTRIBUTOR's ownership of the Distribution Rights, said ownership to continue until:

(a) transferred, assigned or sold by DISTRIBUTOR or anyone acting on behalf of DISTRIBUTOR (this includes a sale by COMPANY for the account of DISTRIBUTOR in the event of a termination of this Agreement upon the terms defined below), or

(b) COMPANY, for legitimate business reasons unrelated to Section 3.1(c) immediately below, ceases to use distributors to distribute Products in the *Newington 2* market area. In such event, COMPANY will repurchase the Distribution Rights from DISTRIBUTOR at ten (10) times the average weekly sales volume of Products (not Authorized Products) in the Territory calculated over the six (6) month period preceding the repurchase, or

(c) either party exercises its right to terminate pursuant to Section 16.1. In such event, COMPANY will repurchase the Distribution Rights from DISTRIBUTOR at fair market value, which in no event shall exceed the purchase price paid by DISTRIBUTOR for the Distribution Rights

3.2 Nature of Rights: The parties agree that the Distribution Rights sold to DISTRIBUTOR pursuant to the Bill of Sale can be exercised only pursuant to the terms of this Agreement and that any termination of this Agreement other than in accordance with Section 3.1 (b) or (c) requires DISTRIBUTOR, or anyone acting on behalf of DISTRIBUTOR, to sell such Distribution Rights subject to the terms of this Agreement.

IV. SALE OF PRODUCTS AND AUTHORIZED PRODUCTS

4.1 Products and Authorized Products will be sold to DISTRIBUTOR at such terms and prices as established by COMPANY from time to time. Title and risk of loss shall pass to DISTRIBUTOR upon delivery to DISTRIBUTOR as set forth in Section 11.1 below.

V. BEST EFFORTS/DISTRIBUTOR

5.1 Obligations of DISTRIBUTOR: DISTRIBUTOR agrees and covenants to use DISTRIBUTOR's commercially reasonable best efforts to develop and maximize the sale of Products to Outlets within the Territory and service the Territory in accordance with Good Industry Practice as defined above, including servicing all Outlets in accordance with the Outlet's then-established service requirements. DISTRIBUTOR further agrees and covenants to use DISTRIBUTOR's commercially reasonable best efforts to maximize the sale of Authorized Products to those Outlets in the Territory for which such Authorized Products have been produced and to

distribute such Authorized Products to such Outlets in accordance with Good Industry Practice as defined above, including servicing all Outlets in accordance with the Outlet's then-established service requirements. DISTRIBUTOR shall cooperate with COMPANY on its marketing and sales efforts and ensure its employee(s) maintain a clean and neat personal appearance consistent with the professional image customers and the public associate with COMPANY, and customer requirements. DISTRIBUTOR may not sell products in the Territory which are competitive with the Products or DISTRIBUTOR may sell Authorized Products. noncompetitive products, as long as this does not interfere with the distribution of the Products or Authorized Products. **DISTRIBUTOR specifically** acknowledges and agrees that the obligations herein do not reflect control by COMPANY as to the specific details or manner and means of DISTRIBUTOR'S business, but reflect only COMPANY's interests in the results achieved by DISTRIBUTOR, protecting the reputation of the brands, and protecting the business reputation of both DISTRIBUTOR and COMPANY. As set forth in Section 2.10, DISTRIBUTOR must be properly established, organized and at all times in good standing with the appropriate state agency(s) and in compliance with appropriate state law(s).

5.2 Alternate Distribution: Should DISTRIBUTOR believe a certain account or accounts is/are unprofitable to service, DISTRIBUTOR shall provide COMPANY written notice of such belief and the reasons therefor, including a detailed financial analysis. Within ten (10) business days thereafter, COMPANY representative(s) and DISTRIBUTOR

shall meet to discuss DISTRIBUTOR's concerns. DISTRIBUTOR and COMPANY agree to work in good faith to explore opportunities to remedy the unprofitability of such account(s), and DISTRIBUTOR further agrees to give its best effort to remedy the unprofitability of such account(s), including the exercise of any recommendations made by COMPANY. After exhausting all such efforts, if the COMPANY agrees with DISTRIBUTOR's position, DISTRIBUTOR shall be relieved of its contractual obligation to service such account(s) for a period of time determined by COMPANY. During such period, COMPANY may make alternate distribution arrangements for such account(s) and DISTRIBUTOR shall not receive any credit for sales associated with such alternate distribution. At the end of the period specified by COMPANY, COMPANY and DISTRIBUTOR shall meet again to determine future service of such account(s), including DISTRIBUTOR resuming service once such account(s) become profitable.

5.3 *Non-Compliance*: Failure to comply with Section 5.1 and/or any of the terms, conditions and obligations elsewhere in this Agreement shall be considered a material breach of this Agreement and shall be governed by the provisions of this Agreement dealing with termination.

VI. BEST EFFORTS/COMPANY

6.1 *Obligations of Company*: COMPANY shall use its commercially reasonable best efforts to manufacture and deliver to DISTRIBUTOR sufficient quantities of Products and Authorized Products to supply Outlets requesting service in the Territory, preserve and develop the quality and marketability of the

Products, and cooperate with DISTRIBUTOR on DISTRIBUTOR's sales and marketing efforts. In order to assist DISTRIBUTOR with product sales to Chains, which may require COMPANY and/or its affiliate(s) involvement to obtain authorization to sell Products or Authorized Products, DISTRIBUTOR hereby designates COMPANY and/or its affiliate(s) and COMPANY and/or its affiliate(s) agree(s) to act as DISTRIBUTOR's limited agent to obtain such authorization. This limited agency designation is for the sole purpose of authorization discussions (including space, position and pricing) with regard to Chain accounts. Nothing herein shall obligate COMPANY and/or its affiliate(s) seek to authorization or otherwise pursue business opportunities with any specific Chain or other account. Additionally, nothing herein shall prevent DISTRIBUTOR from having the right to deal directly with Chain accounts with regard to such terms, although DISTRIBUTOR acknowledges Chains may decide not to deal with either COMPANY or DISTRIBUTOR.

6.2 Non-Compliance: Failure to comply with Section 6.1 shall be considered a material breach of this Agreement and shall be governed by Section 20.2 below (Company Breach). Repeated breaches by COMPANY which threaten to do substantial harm to DISTRIBUTOR'S business shall entitle DISTRIBUTOR to terminate this Agreement.

VII. ACCOUNTS AND/OR TRADEMARKS EXEMPT FROM DISTRIBUTORSHIP

7.1 *Accounts Exempt From Agreement*: COMPANY reserves the right to solicit and service drop delivery and other non-DSD accounts with Products and

Authorized Products, in whole or in part, in DISTRIBUTOR'S Territory, so long as it is done for legitimate business reasons and not for the purpose of undermining DISTRIBUTOR's business.

- **7.2** *Thrift Stores*: COMPANY reserves the right to continue to sell Products and Authorized Products through its retail thrift store operation, COMPANY-owned and/or independently owned.
- 7.3 Other Trademarks: The parties hereto stipulate that COMPANY and COMPANY affiliates produce a variety of products marketed under a variety of trademarks distributed through multiple channels of distribution, including warehouse distribution. This Agreement does not restrict any other distribution of products by COMPANY and COMPANY affiliates marketed under trademarks not listed on **Exhibit B** via warehouse distribution or otherwise.

VIII. PAYMENT FOR PRODUCTS

- 8.1 Settlement of Account: On or before Friday of each week, DISTRIBUTOR will remit to COMPANY a full settlement of all Products and Authorized Products sold to DISTRIBUTOR in the preceding week, in accordance with terms established by COMPANY from time to time. DISTRIBUTOR authorizes COMPANY to make credits and charges to its weekly settlement as set forth on the Settlement Statement Authorization, a sample of which is attached hereto as Exhibit G.
- 8.2 *Cash Sales*: DISTRIBUTOR is solely responsible for the collection of all cash sales in the Territory.
- 8.3 *Non-Cash Sales*: In cases where Products and/or Authorized Products are sold and distributed to Outlets which have been approved by COMPANY for

credit and in its sole discretion, COMPANY will accept electronic data with corresponding proof of delivery, charge slips, Outlet-generated authorizations or other forms of payment authorizations as may be required by the Outlet or (including but not COMPANY limited to. authorizations signed by the Outlet's authorized representative, such as the store manager, or its designee) in lieu of cash, and credit DISTRIBUTOR'S account for such sales, provided DISTRIBUTOR fully complies with COMPANY'S credit policies as established from time to time. In the event DISTRIBUTOR fails to comply with COMPANY's credit policy, fails to provide complete documentation as required by the Outlet or COMPANY, or falsifies any documentation or credit information, COMPANY shall be entitled to charge DISTRIBUTOR's account for any credit extended to therefrom DISTRIBUTOR resulting without limiting any other remedies available to COMPANY. DISTRIBUTOR is wholly responsible for collection of accounts receivable not authorized by COMPANY. COMPANY is obligated to credit DISTRIBUTOR only for such payments made by the Outlet.

- 8.4 *Chain Accounts*: COMPANY and/or its affiliate(s) reserves the right to continue carrying the accounts receivable for all Chain and other major accounts and to promptly credit DISTRIBUTOR for DISTRIBUTOR's sales to all such accounts.
- 8.5 Scan Based Trading: In the event a Chain account implements Scanned Based Trading (SBT), Pay By Scan (PBS) or other accounting methodologies and/or technology, DISTRIBUTOR agrees to comply with policies and procedures as may be

necessary to comply with such Chain accounting and/or technology requirements.

- 8.6 Security Interest: To secure the prompt payment and timely performance of all indebtedness, obligations and liabilities of DISTRIBUTOR to COMPANY under this Agreement, whether now existing or hereafter arising, DISTRIBUTOR hereby grants and conveys to COMPANY a presently existing and continuing security interest in the Distribution Rights, in this Agreement, in all Products and/or Authorized Products now or hereafter in DISTRIBUTOR's possession, in all accounts now or hereafter arising out of the sale by DISTRIBUTOR of Products and/or Authorized proceeds thereof. Products. and in all DISTRIBUTOR agrees that this security interest attaches immediately upon execution of this Agreement by DISTRIBUTOR and that COMPANY has all of the rights of a secured party under the applicable Uniform Commercial Code, as amended from time to time. DISTRIBUTOR expressly authorizes COMPANY to file and refile all appropriate Uniform Commercial Code financing statements necessary to perfect the security interests granted hereunder.
- 8.7 **Default**: Nothing herein shall be deemed to require COMPANY to fill an order of DISTRIBUTOR during the time when DISTRIBUTOR has failed to make any payment due to COMPANY in a timely fashion.

IX. EQUIPMENT AND INSURANCE

9.1 *Delivery Vehicles*: DISTRIBUTOR is responsible for obtaining DISTRIBUTOR's own delivery vehicle(s) and purchasing adequate insurance

described Section 9.2 below. thereon, as in good Consistent with industry practice. DISTRIBUTOR agrees to keep the delivery vehicle(s) clean at all times and in a manner consistent with the professional image customers and the public associates with COMPANY, and to maintain the delivery vehicle(s) in such condition as to provide safe, prompt, and regular service to all customers. If legally required, DISTRIBUTOR agrees to have painted in a conspicuous manner on any delivery vehicle owned or leased by it to carry out the terms hereof: "Owned and Operated by name), (DISTRIBUTOR's An Independent Contractor".

- 9.2 Insurance: DISTRIBUTOR shall maintain at all times throughout the duration of this Agreement insurance policies from insurance companies that maintain a financial strength rating of A- or better and are licensed to do business in the state(s) in which the Territory is located, and naming COMPANY as an additional insured party on all such insurance policies. DISTRIBUTOR may select either option as set forth in Exhibit H attached hereto and made a part hereof. DISTRIBUTOR agrees to provide COMPANY with written evidence of insurance and endorsement within three (3) days upon request by COMPANY. COMPANY reserves the right to change the insurance coverage requirements by providing DISTRIBUTOR with thirty (30) calendar days notice.
- **9.3** Other Insurance: In addition to the insurance coverage required under this Agreement, it is DISTRIBUTOR'S responsibility to carry and

maintain any other insurance coverage DISTRIBUTOR may desire.

X. PROPRIETARY SERVICES

- 10.1 Services Provided: COMPANY shall make available, and DISTRIBUTOR shall use, certain proprietary administrative services in order to assist DISTRIBUTOR in the conduct of DISTRIBUTOR's business for the following purposes: (i) collection of data; (ii) preparation of sales sales tickets: (iii) accumulation of sales histories; (iv) preparation of daily and weekly settlements; (v) preparation of "adds" to "cuts" automated and from DISTRIBUTOR's daily order of Products and Authorized Products; (vi) direct communication to COMPANY for the purpose of DISTRIBUTOR'S ordering of product and receipt of daily load information; (vii) providing automated route book information; (viii) providing automatic product movement information; (ix) providing individual customer sales profiles; and (x) providing suggested orders for each customer. DISTRIBUTOR specifically acknowledges and agrees that the use of such proprietary administrative services is a necessary component of transacting business with Outlets and maximizing sales and does not reflect control by COMPANY of the specific details or manner and means of DISTRIBUTOR's business.
- **10.2** *Administrative Fee*: COMPANY shall charge, and DISTRIBUTOR shall pay, a fair market, reasonable administrative fee for these services, which shall be established from time to time by COMPANY.
- **10.3** *Confidentiality*: DISTRIBUTOR agrees to maintain and hold in confidence any and all

information obtained from or derived about COMPANY's proprietary administrative services pursuant to the provisions of Section 20.7.

- **10.4** *No Proprietary Rights*: DISTRIBUTOR acknowledges that DISTRIBUTOR cannot and shall not acquire any proprietary rights in COMPANY's proprietary administrative services and no rights shall accrue to DISTRIBUTOR by virtue of the use of the proprietary administrative services. The right to use these proprietary administrative services shall not be assigned by DISTRIBUTOR without the express written consent of COMPANY.
- 10.5 Termination or Amendment of Services: This provision shall remain in effect for as long as this Agreement is in effect. However, in the event this Agreement is terminated for any reason, so shall DISTRIBUTOR'S right to use the proprietary administrative services. COMPANY may, in its sole discretion, alter, amend or terminate the proprietary administrative services herein made available by the giving of fourteen (14) calendar days' notice to DISTRIBUTOR.
- 10.6 Accurate Truthful and **Reporting**: DISTRIBUTOR shall at all times utilize the proprietary services in an honest and ethical manner. This specifically includes accurate and truthful reporting of transactions and related information so as to allow COMPANY to comply with applicable legal disclosure reporting requirements, and COMPANY reserves right to audit the DISTRIBUTOR's transactional records and inventory to ensure accurate and truthful reporting and COMPANY's compliance with applicable legal requirements.

XI. PRODUCT DELIVERY

11.1 COMPANY shall deliver Products and Authorized Products to DISTRIBUTOR at the location chosen by DISTRIBUTOR, as reflected on **Exhibit I** attached hereto and made a part hereof. DISTRIBUTOR will be charged a fee for warehouse use. DISTRIBUTOR may request to change the location upon thirty (30) days' notice to COMPANY, provided COMPANY is able to accommodate the requested change, such decision being COMPANY's sole discretion. If COMPANY is not able to accommodate the requested change, it may provide DISTRIBUTOR with other location options.

XII. PRODUCT CODE

- 12.1 Maintaining a fresh market is a fundamental tenet of the baking industry. Accordingly, Out of Code Products or Authorized Products left in the market is a material breach of this Agreement. Repeated violations will be grounds for termination of this DISTRIBUTOR Agreement. specifically acknowledges and agrees that removing Stale Products and Authorized Products from the market on the appropriate color code or designated pick up date is necessary to maintain the reputation of the brands and to protect the business reputation of both DISTRIBUTOR and COMPANY.
- 12.2 DISTRIBUTOR is not obligated to sell its Stale Products and Authorized Products back to COMPANY. However, to assist DISTRIBUTOR in maintaining a fresh market, COMPANY will repurchase a certain percentage of DISTRIBUTOR'S Stale Products or Authorized Products, in accordance with COMPANY's Stale

allowance policy as established by COMPANY from time to time.

12.3 DISTRIBUTOR may not sell any Out of Code Products or Authorized Products, or those not in a saleable condition for distribution to the general public, but may otherwise sell such products to purchasers for non-human consumption.

XIII. ADVERTISING AND PROMOTIONS

- 13.1 COMPANY will provide all COMPANY initiated advertising material at no cost to DISTRIBUTOR, including, but not limited to, point of sale material and COMPANY scheduled media advertising. Subject to COMPANY's prior approval, which approval will not be unreasonably withheld, DISTRIBUTOR may use other advertising materials.
- 13.2 DISTRIBUTOR will adhere to all promotions and feature pricing with respect to the major and Chain accounts in the Territory. Such promotions and feature pricing are optional with respect to local accounts. When DISTRIBUTOR follows promotions or feature pricing, DISTRIBUTOR will receive a reduction in the purchase price accounting for such feature promotion or pricing, such that DISTRIBUTOR and COMPANY share the price allowance. With respect to local accounts, DISTRIBUTOR shall be required to provide COMPANY with sufficient proof that the promotion or feature pricing was extended to the local accounts(s) in order to obtain a pricing adjustment for such sales. DISTRIBUTOR and COMPANY will each share in any resulting pricing allowance(s).

XIV. SERVICE REQUIREMENTS

- 14.1 DISTRIBUTOR is responsible for providing proper service of the Territory at all times.
- 14.2 If DISTRIBUTOR does not service the Territory, for any reason, COMPANY reserves the right to service the Territory, and DISTRIBUTOR agrees to pay a daily fee as indicated on **Exhibit J**, plus any operating expenses COMPANY incurs, including but not limited to fuel, truck rental and travel costs. COMPANY reserves the right to change its dally fee. Such temporary service by COMPANY does not relieve DISTRIBUTOR of the obligation imposed on it by the AGREEMENT nor act to cure any breach by DISTRIBUTOR.

XV. TRANSFER OR SALE OF RIGHTS

15.1 Conditions: The Distribution Rights are owned by the DISTRIBUTOR and may be sold transferred in whole or in part by DISTRIBUTOR, or anyone acting on behalf of DISTRIBUTOR, subject to the prior written approval of COMPANY, which approval shall not be unreasonably withheld. Any sale or transfer of the Distribution Rights must be bona fide, for legitimate business purposes, and compliant with all applicable Additionally, the prospective purchaser or laws. transferee must be fully qualified to meet all of the obligations under this Agreement, including but not limited to, conducting the business as a corporation pursuant to the conditions set forth in Section 2.10. COMPANY'S right of approval in this Section shall expire if not exercised within thirty (30) days after the later of (a) receipt by COMPANY of written notice from DISTRIBUTOR of its intent to sell or transfer to a named bona fide purchaser or

transferee on terms and conditions fully set forth in such notice, and (b) an evaluation by COMPANY of the proposed purchaser or transferee. The transfer, whether in one or a series of transactions, of a majority of the ownership or voting interests in the stock of DISTRIBUTOR, whether by operation of law or otherwise, shall constitute a sale, subject to the prior written approval of COMPANY for purposes of this Article. DISTRIBUTOR shall execute and deliver to COMPANY and its affiliates a release of all its interests and claims to and in such Distribution Rights and all of its interests under or arising out of this Agreement, together with a general release of claims against COMPANY and its affiliates, in the event of any transfer or sale of the Distribution Rights. COMPANY will concurrently execute a general release of claims as to DISTRIBUTOR, except for any claim for monies due and owing COMPANY. Except as specifically set forth in Article III hereof, COMPANY has no obligation to repurchase the Distribution Rights.

- **15.2** Settlement of Account: No transfer or sale of DISTRIBUTOR's right under this Agreement is to be made unless and until DISTRIBUTOR has settled with COMPANY all outstanding accounts and settled all other outstanding liens and debts related to the distributorship; provided, however, that COMPANY may waive this requirement depending on the circumstances of the particular transaction, including if it is satisfied that under the terms of the sale, the purchaser assumes all such liabilities and is financially capable of such assumption.
- 15.3 *Transfer Fee:* In the event of a sale by DISTRIBUTOR, or by anyone other than

COMPANY acting on behalf of DISTRIBUTOR of DISTRIBUTOR's Distribution Rights to a third party, DISTRIBUTOR shall be solely responsible for ensuring the purchaser is fully qualified to operate the business hereunder, and shall pay a transfer fee to COMPANY in an amount equal to two percent (2%) of such gross sales price, but not less than \$2,000, in consideration of the administrative activities undertaken by COMPANY in connection with such sale. In the event of a sale of the Distribution Rights to COMPANY, or by COMPANY behalf on of DISTRIBUTOR, DISTRIBUTOR shalt pay a transfer fee to COMPANY in an amount equal to five percent (5%)of such gross sales price, but not less than \$3,000, in consideration of the administrative activities undertaken by COMPANY in connection with such sale.

15.4 *Taxes* and *Reporting Requirements*: DISTRIBUTOR shall be solely responsible for all taxes and transactional reporting requirements, including but not limited to the nature of the conveyance, related to the transfer or sale of the Distribution Rights.

XVI. INDEPENDENT BUSINESS

16.1 *Essential Term*: The status of DISTRIBUTOR pursuant to this Agreement is that of independent contractor for all purposes and the parties hereby signify their express intention to this effect. DISTRIBUTOR shall not be controlled by COMPANY as to the specific details or manner and means of DISTRIBUTOR's business, it being understood that the interests of COMPANY are the results achieved by DISTRIBUTOR, protecting the reputation of the brands, and protecting the business reputation of both DISTRIBUTOR and COMPANY. DISTRIBUTOR's business is separate and apart from that of COMPANY and it is of the essence of this Agreement that DISTRIBUTOR is an independent business. Any final determination that DISTRIBUTOR is not an independent contractor shall entitle either party to cancel this Agreement, such determination being contrary to the parties' express intention herein to create an independent contractor relationship. Neither DISTRIBUTOR nor any of DISTRIBUTOR's employees, agents, or servants shall be considered or deemed in any way to be employees, agents or servants of COMPANY and neither party has the right or power, express or implied, to do any act or thing that would bind the other, except as herein specifically provided. The parties do not intend to act as joint employers, parent/subsidiary, joint venturers, or any other legal capacity other than separate and distinct businesses acting pursuant to the terms of this Agreement. There is no fiduciary relationship between the parties. Furthermore, none of the benefits provided by COMPANY to its employees are available from DISTRIBUTOR COMPANY to or to DISTRIBUTOR'S employees, agents, or servants. DISTRIBUTOR In the event and/or DISTRIBUTOR's employees, agents or servants hereafter become eligible to participate in any such benefits. DISTRIBUTOR, on behalf of DISTRIBUTOR and DISTRIBUTOR's employees, agents and servants, hereby waives any right to participate in such benefits. Such waiver is not dependent upon DISTRIBUTOR's status as an independent contractor. DISTRIBUTOR will be

solely and entirely responsible for DISTRIBUTOR's acts and for the acts of DISTRIBUTOR's employees, agents, and servants during the performance of this Agreement, and will save and hold COMPANY harmless from any and all damages which may arise therefrom, including attorneys' fees.

- 16.2 Non-Personal Service: This Agreement does not require that DISTRIBUTOR's obligations hereunder be conducted personally, or by any specific individual in DISTRIBUTOR's organization. DISTRIBUTOR shall be free to engage such persons as DISTRIBUTOR deems appropriate to assist in discharging DISTRIBUTOR's responsibilities hereunder. Any breach of this Agreement by any person engaged by DISTRIBUTOR shall be deemed to be a breach by DISTRIBUTOR.
- **16.3 DISTRIBUTOR Employees:** DISTRIBUTOR shall ensure all persons assisting in fulfilling DISTRIBUTOR'S obligations hereunder are treated as employees, not as contractors, for all purposes including, but not limited to, payment for services, payroll and income taxes, tax withholding, and legally required insurance. DISTRIBUTOR agrees to provide proof of such compliance at request of COMPANY.
- **16.4** *Professional Services*: DISTRIBUTOR may engage any legal and/or accounting professional services it deems necessary for purposes of legal compliance, meeting its obligations under this Agreement, and otherwise.

XVII. TERMINATION BY COMPANY

17.1 *Performance*: Except as set forth in Sections 3.1 (b) and (c) and 16.1 above, or this Article, COMPANY

shall not terminate or cancel this Agreement, provided DISTRIBUTOR faithfully carries out the terms hereof. In the event DISTRIBUTOR fails to perform DISTRIBUTOR's obligations under this Agreement, COMPANY may terminate this Agreement as set forth below.

- 17.2 Non-Curable Breach: COMPANY may terminate upon twenty-four (24) hours' written notice and DISTRIBUTOR shall have no right to cure if DISTRIBUTOR's failure of performance Involves criminal activity, threatens public or private health or safety, involves violent activity or violations of law, or threatens to do substantial harm to COMPANY's business, trademarks or reputation, including, but not limited to, any action or inaction on DISTRIBUTOR's part results that in DISTRIBUTOR's inability to service any Chain account. COMPANY may also terminate upon twenty-four (24) hours written notice and DISTRIBUTOR shall have no right to cure if DISTRIBUTOR fails to maintain the insurance coverage requirements as set forth in Section 9.2 above and Exhibit H, or if DISTRIBUTOR dissolves its corporate status.
- 17.3 Curable Breach: In any event of failure of performance by DISTRIBUTOR, COMPANY must give DISTRIBUTOR ten (10) business days written notice within which DISTRIBUTOR may cure DISTRIBUTOR's failure of performance. If DISTRIBUTOR does not cure such failure of performance within this ten (10) day period, COMPANY may thereafter terminate this Agreement and DISTRIBUTOR shall have no further right to cure. Furthermore, the parties agree
that repeated violations, even if cured, constitute a chronic failure of performance and threaten substantial harm to COMPANY's business, trademarks or reputation, and in such event COMPANY shall be entitled to terminate this Agreement immediately and DISTRIBUTOR shall have no further right to cure.

17.4 Actions Following Termination: If this Agreement is terminated under either Section 17.2 or 17.3, COMPANY, within the limits of its ability to do so, will operate the business for the account of DISTRIBUTOR, deducting its reasonable expenses in connection with the operation thereof, and sell DISTRIBUTOR's Distribution Rights to a qualified purchaser(s) at the best price which can reasonably be obtained after proper notice and advertisement. Such sale shall be for the account of the terminated DISTRIBUTOR, and the proceeds of such sale, after deducting therefrom any monies owed bv DISTRIBUTOR to COMPANY, the amount of any outstanding liens, any other known liabilities of the distributorship and the reasonable costs incurred in effecting the sale, shall be turned over to DISTRIBUTOR in exchange for the release of DISTRIBUTOR's Distribution Rights and interests under this Agreement, together with a general release of claims as to COMPANY and its affiliates. COMPANY will concurrently execute a general release of claims as to DISTRIBUTOR, except for any claim for monies due and owing COMPANY.

XVIII. DISPUTE RESOLUTION

18.1 *Negotiation*: DISTRIBUTOR and COMPANY shall attempt in good faith and employ their best efforts to resolve and terminate any controversy, claim or

dispute arising out of or relating to this Agreement promptly by negotiations. Such negotiations shall take place between representatives of COMPANY and DISTRIBUTOR who have the authority to settle and resolve the dispute. Such negotiations shall begin upon written notice from one party to the other describing any dispute or claim which has not been resolved in the ordinary course of business and suggesting a location for a meeting between the parties to conduct such negotiations within ten (10) business days after delivery of such notice. Representatives of DISTRIBUTOR and COMPANY shall meet at a mutually acceptable time and place within ten (10) calendar days after delivery of such notice and thereafter as often as they reasonably deem necessary, to exchange relevant information and to attempt to resolve the dispute. Any such meeting may be attended by one or more representatives of each party. No such meeting shall be attended by an attorney representing either party unless such party shall have first given the other party at least three (3) calendar days' notice that it will be accompanied by an attorney at the next scheduled meeting, and at such meeting the other party may also be accompanied by an attorney.

18.2 *Mediation*: If the dispute has not been resolved within thirty (30) calendar days of the initial notice, or if the party receiving such notice has failed to meet within fifteen (15) calendar days, either party may initiate mediation by a request therefore in writing to the other party. Upon receipt of such notice, DISTRIBUTOR or COMPANY shall be obligated to engage in mediation. If the parties fail to agree within fifteen (15) calendar days of the date of such request for mediation on the selection of a mediator,

then the Center for Public Resources shall appoint a mediator in accordance with its Mediation Procedure and the parties shall continue efforts through mediation to resolve the controversy and dispute between them until mediation is terminated by the occurrence of any of the following events:

- (i) A written resolution in settlement of the dispute is reached, or
- (ii) The mediator informs the parties in writing that further efforts would not be productive or useful, or
- (iii) The parties agree in writing that further efforts would not be productive, or
- (iv) Sixty (60) calendar days elapse from the commencement of the mediation without resolution.

Neither COMPANY nor DISTRIBUTOR may withdraw from mediation before such a termination of the process.

18.3 *Mandatory and Binding Arbitration*: All claims, disputes, and controversies arising out of or in any manner relating to this Agreement or any other agreement executed in connection with this Agreement, or to the performance, interpretation, application or enforcement hereof, including, but not limited to breach hereof and/or termination hereof, which has not been resolved pursuant to the negotiation and mediation provisions herein shall be submitted to binding arbitration in accordance with the terms and conditions set forth in the Arbitration Agreement attached hereto as **Exhibit K**, excepting only such claims, disputes, and controversies as specifically excluded therein.

- **18.4** *Confidentiality*: All negotiations and mediations pursuant to this Article of the Agreement are confidential and shall be treated as compromise and settlement negotiations for purposes of the Federal Rules of Evidence and any similar state rules of evidence.
- 18.5 *Tolling*: All applicable statutes of limitation and defenses based upon the passage of time shall be tolled during the negotiation and mediation procedures specified in this Article. All deadlines specified in such procedures may be extended by mutual agreement of the parties.

XIX. TRADEMARKS, TRADE NAMES, AND PROPRIETARY MATERIALS

19.1 Permission for Use: Subject to the terms and conditions of this Article and to the Distribution Rights granted herein, for the Duration of this Agreement, COMPANY authorizes DISTRIBUTOR to use, on a non-exclusive basis, the trade names and trademarks (collectively, "Trademarks"), trade dress, package designs, logos, artwork, advertising materials and other works protected by copyright, and other intellectual property of or associated with the Products (collectively, including the Trademarks, the "Proprietary Materials") solely in connection with DISTRIBUTOR's advertising, promoting, marketing, sale, and distribution of Products in the DISTRIBUTOR may not assign or Territory. transfer this authorization except in the context of a sale or transfer of all or a portion of the Distribution provided in this Rights as Agreement. DISTRIBUTOR will not use any of the Proprietary Materials in any manner different from how they are used generally by COMPANY.

19.2 Conditions of Use: DISTRIBUTOR shall make no use of the Trademarks or Proprietary Materials, nor engage in any program or activity which makes use of or contains any reference to COMPANY, its products, Trademarks or Proprietary Materials, other than as specifically provided herein, except with the prior written consent of COMPANY. Should DISTRIBUTOR advertise other services or products on its delivery vehicle, such advertising shall not dilute or negatively affect the reputation and goodwill of COMPANY, the Trademarks or the Proprietary Materials. DISTRIBUTOR will not (i) do anything that would or might be inconsistent with, impair, or conflict with ownership of the Proprietary Materials by COMPANY or, if licensed, franchised, or otherwise, the owner thereof; (ii) use any of the Proprietary Materials in any manner likely to (a) deceive or mislead the public, (b) endanger the validity or enforceability of any of the Proprietary Materials, or (c) damage or impair the reputation or value of COMPANY, any of the Proprietary Materials, or any Product; (iii) attack, contest, oppose, or interfere with the right, title, or interest in or to any Proprietary Materials of COMPANY or the owner thereof in any jurisdiction; or (iv) adopt or use any work confusingly or substantially similar to any of the Proprietary Materials. DISTRIBUTOR will not use any of the Trademarks or Proprietary Materials in either its legal or fictitious trade name without the prior written consent of COMPANY. DISTRIBUTOR will not use any of the Trademarks, Proprietary Materials, or any variation thereof as a business and/or corporate name or d/b/a of DISTRIBUTOR, an Internet domain name or a mnemonic, cipher, or "vanity" telephone number without COMPANY's prior written approval.

- **19.3** *Monitoring and Cooperation*: To insure the uniform quality of the goods in connection with which the Proprietary Materials are used, COMPANY will have the right to require DISTRIBUTOR to immediately discontinue any non-approved use of any Proprietary Materials or any other activity that COMPANY in its sole judgment considers likely to impair the value or validity of the Proprietary DISTRIBUTOR will immediately Materials. observe and require that other persons under its control observe any and all instructions of the COMPANY limiting or halting the use of the Proprietary Materials. DISTRIBUTOR will notify COMPANY, in writing, as soon as DISTRIBUTOR becomes aware of any infringement, imitation, impairment, counterfeiting, or other unlawful or improper use of any of the Proprietary Materials or any of the Products. DISTRIBUTOR will cooperate with and provide testimony and other evidence to COMPANY in and in connection with any application for, registration and protection of, or enforcement of the Proprietary Materials in the Territory as and when requested by COMPANY.
- **19.4** *Acknowledgment*: DISTRIBUTOR acknowledges and agrees that (i) all the rights concerning the Proprietary Materials are owned by COMPANY or the owner thereof, (ii) DISTRIBUTOR does not have, and will not as a result of this Agreement or performance hereunder acquire, any ownership or legal rights in any of the Proprietary Materials used in connection with the Products, and

(iii) DISTRIBUTOR's sole rights with respect to the Proprietary Materials are set forth in this Article.

19.5 *Return Upon Termination*: Upon termination of this Agreement, DISTRIBUTOR shall cease use of COMPANY's Proprietary Materials and shall deliver to COMPANY any advertising, promotional or merchandising material.

XX. MISCELLANEOUS

- 20.1 Notice: Any notice required or permitted under this Agreement shall be deemed properly given and received on the date received if hand delivered; on the next business day after deposit with a nationally recognized overnight courier service for overnight delivery; or three (3) days after deposited in the mails, return receipt requested, registered or certified mail postage pre-paid, addressed to the parties at the addresses first referenced above, or at such other address as either party may designate by written notice given pursuant to this section.
- 20.2 Company Breach: If DISTRIBUTOR maintains that COMPANY is in breach of this Agreement, DISTRIBUTOR should COMPANY notify President in writing, by certified mail, return receipt requested. Such written notice must include the specific Section of this Agreement DISTRIBUTOR maintains has been breached and sufficient facts to provide reasonable notice to COMPANY of the action or failure to act which DISTRIBUTOR maintains is a breach of this Agreement. Upon receipt of the DISTRIBUTOR'S notice of breach, COMPANY shall have a reasonable period of time to investigate and cure any breach. Any failure to comply with this provision shall not affect

DISTRIBUTOR's right to submit a claim to mandatory and binding arbitration.

- **20.3** *Survival*: This agreement shall be binding upon the heirs, personal representatives, successors or assigns of the parties to this Agreement. Notwithstanding any termination of this Agreement, all provisions hereof that, by their terms or reasonable interpretation thereof, set forth obligations that extend beyond the termination of this Agreement, shall survive and remain in full force and effect.
- **20.4** *Incorporation of Bill of Sale*: This Agreement is subject to and affected by a Bill of Sale simultaneously executed by the parties, such Bill of Sale being fully incorporated in this Agreement by reference.
- 20.5 *Entire Agreement*: This Agreement, together with all Exhibits incorporated herein by reference, as amended from time to time in accordance with the Agreement, sets forth the entire agreement between the parties and supersedes all prior agreements, discussions, negotiations, understandings, representations, conditions, warranties and covenants between them with respect to this subject matter. Unless set forth in this Agreement, no party shall be liable for any representation made to any other. Except as otherwise expressly authorized herein, this Agreement may be amended or modified only by a writing signed by all parties.
- **20.6** *Indemnification*: DISTRIBUTOR agrees to indemnify and hold harmless COMPANY from and against any and all claims, actions, liabilities, expenses, losses or demands, including reasonable attorneys' fees, growing out of or based upon this

Agreement acts omissions of or any or DISTRIBUTOR arising in the normal course of the conduct of business by DISTRIBUTOR pursuant to DISTRIBUTOR shall also this Agreement. Indemnify COMPANY against all liability and loss in connection with, and shall assume full responsibility for, payment of all federal, state and local taxes and/or contributions imposed or required under unemployment insurance, workers' compensation insurance, and income tax laws with respect to DISTRIBUTOR or DISTRIBUTOR's employees engaged in performance of this Agreement.

20.7 Trade Secrets/Confidential Business Information: DISTRIBUTOR recognizes and acknowledges that COMPANY and its affiliated entitles have, through the expenditure of substantial time, effort and money, developed and acquired certain trade secrets and confidential business information ("Confidential Information") which are of great value to COMPANY. DISTRIBUTOR further acknowledges and understands that through the distributor relationship with COMPANY, DISTRIBUTOR will access to Confidential Information have of COMPANY and its affiliated entities. Accordingly, except as and to the extent required by law, DISTRIBUTOR covenants that DISTRIBUTOR will make no disclosure, or any use, in any manner whatsoever, of any of the Confidential Information of COMPANY and/or any of its affiliated entities while this Agreement is in effect and thereafter, except as specifically authorized by COMPANY. Upon termination of this Agreement, DISTRIBUTOR shall return to COMPANY all embodiments of such Confidential Information that are in DISTRIBUTOR'S possession, custody or control.

As used herein the term Confidential Information includes (a) sales data, financial statements and information, marketing arrangements and plans, trade secrets, pricing information and strategies, business development plans, and any other business methods, processes and techniques of COMPANY and/or its related entities to the extent all of which are not generally known to the trade or industry and/or (b) confidential or proprietary information or trade secrets of third parties, including retail customers, with which COMPANY and/or its related entities conducts business and which is supplied to DISTRIBUTOR for purposes of this Agreement. Notwithstanding the foregoing, Confidential information shall not include information that: (i) is or becomes generally known to the public not as a result of a disclosure by DISTRIBUTOR; (ii) is received by DISTRIBUTOR in good faith and without restriction from a third party having the right to make such disclosure; and/or (iii) is known to DISTRIBUTOR at the time of COMPANY's disclosure thereof to DISTRIBUTOR or DISTRIBUTOR's becoming aware thereof in the course of and as a result of DISTRIBUTOR's business relationship with COMPANY. DISTRIBUTOR's obligations in this paragraph shall be in addition to, and not in lieu of, any and all obligations imposed upon DISTRIBUTOR by applicable law.

20.8 *Injunctive Relief*: DISTRIBUTOR agrees and acknowledges that a violation of Section 20.7 will cause irreparable harm to COMPANY and that there will be no adequate remedy at law to redress the resulting harm that will be suffered by COMPANY. Therefore, DISTRIBUTOR further agrees that in

the event of any violation or threatened violation of this Section, COMPANY shall be entitled as a matter of course to an immediate injunction out of any court of competent jurisdiction restraining such violation or threatened violation by DISTRIBUTOR, such right to be cumulative and in addition to whatever other remedies, at law or in equity, that COMPANY may have.

- 20.9 Savings Provision/Non-Waiver: Should any portion, word, clause, sentence or paragraph of this Agreement be declared void or unenforceable, including the Arbitration Agreement attached hereto, such portions shall be modified or deleted in such a manner as to make this Agreement as modified legal and enforceable to the fullest extent permitted under applicable law. Except as expressly provided herein, no waiver by either party of any default in the performance of any part of this Agreement by the other party shall be deemed a waiver of any other default hereunder.
- **20.10** *Legal Counsel*: DISTRIBUTOR acknowledges that it has been given a reasonable opportunity to discuss this Agreement with an attorney of its choosing; that it has carefully read and fully understands the provisions of this Agreement, and that it is entering into this Agreement knowingly, voluntarily, and with the intention of being legally bound by all of the terms of this Agreement.
- **20.11** *Governing Law*: This Agreement and the construction thereof shall be governed by the laws of the State of Connecticut, without regard to the conflict-of-law rules.
- 20.12 *Revocation*: This Agreement, including the Arbitration Agreement attached hereto, may be

revoked by DISTRIBUTOR within seven (7) calendar days after DISTRIBUTOR executes it ("Revocation Period") by sending written notice of revocation to Peter Roy, and the Agreement will not become effective or enforceable until this Revocation Period has expired, without revocation.

IN WITNESS WHEREOF, COMPANY and DISTRIBUTOR have executed this Agreement this 7th day of June, 2017, and each signatory hereto represents that he or she has the authority to execute this Agreement and legally bind the respective party.

DISTE	RIBUTOR	COMF	COMPANY				
s/Neal	Bissonnette	s/Mike Brock					
By:	Neal Bissonnette	By:	Mike Brock				
Its:	President	Its:	Vice President				

WITNESSES:

ATTEST:

EXHIBIT "A" — TERRITORY

Territory begins at intersection of Northfield Road and Route 6. Go north on Route 6 to Hill Street. Turn left on Hill Street to James P. Casey Road. Turn right on James P. Casey Road and James P. Casey Road turns into Peacedale Street. Continue on Peacedale Street and Peacedale Street turns into Maple Ave. Continue on Maple Ave and Maple Ave turns into Mix Street. Continue on Mix Street to Route 6. Turn left on Route 6 to New Britain Ave. Turn left on New Britain Ave to Meadow Road. Turn right on Meadow Road to Route 10. Turn right on Route 10 to Northwest Drive. Turn right on Northwest Drive to Route 177. Turn left on Route 171 to West Main Street. Turn right on West Main Street and West Main Street turns into Pine Street. Continue on Pine Street to Route 72. Turn left on Route 72 and Route 72 turns into Mountain Road. Continue and Mountain Road and Mountain Road turns into South Street. Continue on South Street to West Street. Turn right on West Street to Route 72. Turn left on Route 72 to Route 6 working both sides throughout.

Acknowledged and Agreed:

s/Neal Bissonnette DISTRIBUTOR s/Mike Brock COMPANY

EXHIBIT "B" – PRODUCTS

Except as expressly limited by the Distributor Agreement, DISTRIBUTOR shall have the exclusive right to sell and distribute the following Products in the Territory specifically described in **Exhibit A**.

- 1. Nature's Own
- 2. Country Kitchen
- 3. Barowsky's
- 4. TastyKake
- 5. Wonder Bread
- 6. Home Pride
- 7. Cobblestone Bread Company
- 8. Old Fashion Hearth

EXHIBIT "C" — AUTHORIZED PRODUCTS

Except as expressly limited by the Distributor Agreement, DISTRIBUTOR is authorized to distribute the following products. Authorized Products may include products with logos and/or labels, or without. DISTRIBUTOR does not acquire any proprietary or ownership rights to such Authorized Products, exclusive or non-exclusive, and COMPANY may cease selling such Authorized Products to DISTRIBUTOR upon notice. COMPANY may also add Authorized Products to this Exhibit upon notice to DISTRIBUTOR.

- 1. Best Yet
- 2. C&S
- 3. Clear Bag
- 4. HANNAFORD
- 5. Middle East
- 6. Shurfine
- 7. Sunbeam
- 8. SuperBuy
- 9. Great Value
- 10. Dave's Killer Bread Products

EXHIBIT "D" — BILL OF SALE

<u>CK SALES CO., LLC</u>, ("COMPANY"), subject to the consideration set forth below, effective the <u>19th</u> day of <u>June, 2017</u>, hereby conveys, sells, transfers and delivers to <u>Bissonnette Inc</u>, ("DISTRIBUTOR"), the Distribution Rights specifically described in the Distributor Agreement executed simultaneously with this Bill of Sale. Both the Distributor Agreement and all appendices (Exhibits) thereto are specifically incorporated into and made a part of this Bill of Sale.

The terms of this sale are as follows:

Purchase price shall be \$60,650.00. A minimum five percent (5%) down payment shall be required at or before the effective date of the Distributor Agreement.

DISTRIBUTOR shall own all Distribution Rights in the Territory consistent with the terms and conditions contained in the Distributor Agreement.

COMPANY has lawful title to the Distribution Rights in the Territory free from all encumbrances. COMPANY warrants and will defend the rights conveyed herein against the lawful claims and demands of all persons.

IN WITNESS WHEREOF, COMPANY and DISTRIBUTOR execute this Bill of Sale this <u>7th</u> day of **June**, 2017.

DIST	RIBUTOR	COMPANY				
s/Ne	al Bissonnette	s/Mike Brock				
By:	Neal Bissonnette	By:	Mike Brock			
Its:	President	Its:	Vice President			

WITNESSES:

EXHIBIT "E" — OWNER

Owner's Name:Neal P. BissonnetteAddress:47 Harvest Ln

City, State, ZIP Bristol, CT 06010

EXHIBIT "F" — PERSONAL GUARANTY

In consideration of the foregoing premises, covenants and conditions, and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged. Neal P. Bissonnette, an individual ("GUARANTOR"), does hereby irrevocably and unconditionally guarantee to COMPANY, and become surety to COMPANY, for the performance and compliance with the terms, conditions and obligations of this Distributor Agreement entered into by and between Bissonnette Inc (Corporation Name), a Connecticut corporation ("DISTRIBUTOR"), and COMPANY. If any term, condition or obligation of the Distributor Agreement is not complied with, performed, or paid by DISTRIBUTOR as required therein, including any terms, conditions or obligations due following termination thereof. GUARANTOR will, upon COMPANY's demand, immediately ensure the timely and complete performance of DISTRIBUTOR of each and every obligation and duty imposed on it by the Distributor Agreement, and/or pay any amounts due and owing due to DISTRIBUTOR's breach of the Distributor Agreement or as required by the terms and conditions thereof. GUARANTOR shall also pay to COMPANY upon demand all costs and expenses. including but not limited to reasonable attorney fees, which may be incurred by COMPANY in the enforcement collection of monies due and/or hereunder. GUARANTOR is fully responsible for ensuring DISTRIBUTOR is properly established, organized and at all times in good standing with the appropriate state agency(s), and in compliance with appropriate state law(s) in accordance with Section 2.10 of the Distributor Agreement. GUARANTOR agrees and acknowledges he/she is subject to the Arbitration Agreement attached hereto as Exhibit K.

s/Neal Bissonnette	6/7/17
GUARANTOR Signature	Date

Neal Bissonnette

GUARANTOR Name (Print)

54

EXHIBIT "G" — SETTLEMENT STATEMENT AUTHORIZATION

I understand that on a weekly basis I will receive a settlement statement from the Company (a sample of the current format is shown below and is subject to change). I also understand that this statement reflects an ongoing accounting between the Company and me for product purchases, adjustment to product purchases, various credits, and business expenses.

I hereby authorize the Company to make the adjustments noted on the settlement statement, both charges and credits, on a weekly basis. I understand that my portion of customer price allowances is included in the standard discount.

I also authorize the Company to collect from me, via the weekly settlement statement, business expenses such as those shown below, on a weekly basis. I understand and agree that should I dispute any of the credits or charges on a weekly statement that I must bring such dispute to the Company's attention in writing within ten (10) calendar days of receipt of the statement with the disputed credit or charge. If I do not, I understand that my failure to do so indicates my agreement that all charges and credits on the statement are proper.

WITNESS

Ja Dite



Scan Based Trading Adj. to 12/26/2015) Point of Sale Data	Standard Discount Additional Discount Contest Earnings Other Earnings Authorized Charges AR Ticket Chargebacks	Damaged Store Code Product Transfers Cust Price Allwnce/Overrides Stales
(12/20/2015 7,282.12-	4,032.40- 0.00	0.00 0.00 191.69- 4,455.26- 1,866.04-

2,628.33-0.000.000.00

57

License, r ees Truck/Bus Insurance	Truck Repair Truck Tax, Tag,	Truck Rental	Truck Lease	Business Expenses: Stale Adjustment Territory Payment Truck Purchase	Revaluation Variance Shrink New Distributor Balance
					67.33- 0.00 1,103.96
73.79	0.00		182.69	0.00 339.73 0.00	
73.79 Apparel	Open Route Expense Supplies / Equipment	Adjustment Sales and Use Tax	Wk FICA Tax-	Administrative Fee Warehouse Fee FICA Tax-Current	2,628.53-
0.00	0.00	0.00	0.00	36.00 36.00 143.31	1,524.57-

To be paid via Direct Deposit to bank account on record this Friday	Weekly Distributor Balance				Less Current Week SBT Delivery Tickets (12/27/2015 to 01/02/2016):		Less Prior Week Scan Based Trading Inventory (12/26/2015):	Scan Based Trading Credits		Total Distributor Balance	Expense	Total Business	Expense	Life Insurance 0.00 Miscellaneous 0.00	arges	Health/Disability 24.86 Temp Svce Fee 0
4- is Friday	1,025.5	6-	6.5	57	7,	0-	8,058.2		12	14,609.		726.96		-	0.00	0.00

EXHIBIT "H" — DISTRIBUTOR CHOICE OF INSURANCE COVERAGES

Option 1:

- (i) Automobile liability insurance, including hired auto and non-owned auto, on all vehicles used in the business, in such amounts as may reasonably be established by COMPANY from time to time, which as of the date of this Agreement will include combined single limit coverage of at least \$1,000,000; and collision and comprehensive loss coverage for the actual cash or replacement cost value of any and all delivery vehicles operated by DISTRIBUTOR that are financed or leased, with no more than a \$500 deductible. Such coverage shall be primary and not contributory to any other coverage available to COMPANY;
- (ii) Comprehensive general liability coverage, in such amounts as may reasonably be established by COMPANY from time to time, which as of the date of this Agreement will include an occurrence and aggregate limit of at least \$1,000,000. Such coverage shall be primary and not contributory to any other coverage available to COMPANY;
- (iii) Umbrella or excess liability coverage, ln such amounts as may reasonably be established by COMPANY from time to time, which as of the date of this Agreement will include an occurrence and aggregate limit of at least \$1,000,000. Such coverage shall be primary and not contributory to any other coverage available to COMPANY;
- (iv) Workers compensation coverage for any and all employees of DISTRIBUTOR in compliance with all

requirements and laws of the state or states in which DISTRIBUTOR operates.

Option 2:

- (i) Automobile liability insurance, including hired auto and non-owned auto, on all vehicles used in the business, in such amounts as may reasonably be established by COMPANY from time to time, which as of the date of this Agreement will include combined single limit coverage of at least \$2,000,000; and collision and comprehensive loss coverage for the actual cash or replacement cost value of any and all delivery vehicles operated by DISTRIBUTOR that are financed or leased, with no more than a \$500 deductible. Such coverage shall be primary and not contributory to any other coverage available to COMPANY;
- (ii) Comprehensive general liability coverage, in such amounts as may reasonably be established by COMPANY from time to time, which as of the date of this Agreement will include an occurrence and aggregate limit coverage of at least \$2,000,000. Such coverage shall be primary and not contributory to any other coverage available to COMPANY;
- (iii) Workers compensation coverage for any and all employees of DISTRIBUTOR in compliance with all requirements and laws of the state or states in which DISTRIBUTOR operates.

EXHIBIT "I" — DISTRIBUTOR CHOICE OF DELIVERY LOCATION

As set forth in Section 11.1, COMPANY will deliver Products and Authorized Products to DISTRIBUTOR at the location selected below by DISTRIBUTOR.

Please select choice below by initialing the appropriate selection:

1. Location #1 Bridgeport

NB 2. Location #2 Waterbury

3. Location #3 Windsor

EXHIBIT "J" — DISTRIBUTOR CHOICE OF COMPANY FEE FOR SERVICE

As set forth in Section 14.2, if COMPANY services DISTRIBUTOR's Territory, DISTRIBUTOR agrees to pay a daily fee to COMPANY as selected below by DISTRIBUTOR.

Please select choice below by initialing the appropriate selection:

	Option 1:	<i>\$350</i> per day.
NB	Option 2:	12% of average daily net sales.

EXHIBIT "K" — ARBITRATION AGREEMENT

The parties agree that any claim, dispute, and/or controversy except as specifically excluded herein, that either DISTRIBUTOR (including its owner or owners) may have against COMPANY (and/or its affiliated companies and its and/or their directors, officers, managers, employees, and agents and their successors and assigns) or that COMPANY may have against DISTRIBUTOR (or its owners, directors, officers, managers, employees, and agents), arising from, related to, or having any relationship or connection whatsoever Distributor with the Agreement between DISTRIBUTOR and COMPANY ("Agreement"), including the termination of the Agreement, services provided to COMPANY by DISTRIBUTOR, or any other association that DISTRIBUTOR may have with COMPANY ("Covered Claims") shall be submitted to and determined exclusively by binding arbitration under the Federal Arbitration Act (9 U.S.C. §§ 1, et seq.) ("FAA") in conformity with the Commercial Arbitration Rules of the American Arbitration Association ("AAA" or "AAA Rules"), or any successor rules, except as otherwise agreed to by the parties and/or specified herein. Such arbitration shall be conducted before a single arbitrator unless all parties to the arbitration agree otherwise in writing. Copies of AAA's Rules are available on AAA's website (www.adr.org).

COMPANY shall pay for all arbitration filing fees and costs that are customarily associated with AAA arbitration, subject to the Arbitrator's authority to award fees and costs to COMPANY as the prevailing party. Each party may be represented by legal counsel of their own choosing. Each party shall pay its own attorneys' fees, provided that an Arbitrator may award attorney's

fees and costs to the prevailing party under any applicable statute or written agreement to the same extent that attorney's fees and costs could be awarded in court. The arbitration shall be subject to the same burdens of proof and statutes of limitations as if the Covered Claims were being heard in court. The Arbitrator shall issue a written decision within forty-five (45) days of the later of: (1) the arbitration hearing; or (2) submission of the parties' postarbitration briefs. The Arbitrator's written decision shall include findings of fact and conclusions of law. The Arbitrator shall have the authority to award the same damages and other relief that would have been available in court pursuant to applicable law had the Covered Claim been brought on an individual basis in such forum, including attorneys' fees and costs. Subject to the parties' right to appeal, the decision of the arbitrator will be final and binding. The Arbitrator shall not have the authority to add to, amend, or modify, existing law. No arbitration award or decision will have any preclusive effect as to any issues or claims in any dispute, arbitration, or court proceeding where any party was not a named party in the arbitration.

All Covered Claims against COMPANY must be brought by DISTRIBUTOR on an individual basis only and not as a plaintiff or class member in any purported class, collective, representative, or multi-plaintiff action. DISTRIBUTOR further agrees that if it is within any such class, collective, representative, or multi-plaintiff action, it will take all steps necessary to opt-out of the action or refrain from opting in or joining, as the case may be, and DISTRIBUTOR expressly waives any right to recover any relief from any such class, collective, representative, or multi-plaintiff action. Similarly, all Covered Claims by COMPANY against DISTRIBUTOR may not be brought as a plaintiff or class member in any purported class, collective, representative, or multiplaintiff action. The parties understand that there is no right or authority for any Covered Claim to be heard or arbitrated on a multi-plaintiff, collective, or class action basis, as a private attorney general, or any other representative basis. The parties understand that there are no bench or jury trials and no class, collective, representative, or multi-plaintiff actions are permitted under this Arbitration Agreement. The Arbitrator shall not consolidate claims of different distributors into one proceeding, nor shall the Arbitrator have the power or authority to hear arbitration as a class, collective, representative, or multi-plaintiff action. The Arbitrator may award damages on an individual basis only.

TO THE MAXIMUM EXTENT PERMITTED BY LAW, BOTH PARTIES EXPLICITLY WAIVE ANY **RIGHT TO:** (1) INITIATE OR MAINTAIN ANY COVERED CLAIM ON A CLASS, COLLECTIVE, **REPRESENTATIVE**, OR **MULTI-PLAINTIFF** BASIS EITHER IN COURT OR ARBITRATION: PARTICIPATE OR AS (2) SERVE А REPRESENTATIVE ANY OF SUCH CLASS, COLLECTIVE, OR REPRESENTATIVE ACTION; (3) SERVE OR PARTICIPATE AS A MEMBER OF ANY SUCH CLASS, COLLECTIVE, OR **REPRESENTATIVE ACTION; OR (4) RECOVER** FROM ANY SUCH ANY RELIEF CLASS, COLLECTIVE, REPRESENTATIVE, OR MULTI-PLAINTIFF ACTION.

Any dispute concerning the validity or enforceability of this prohibition against class, collective, representative, or multi-plaintiff action arbitration shall be decided by a court of competent jurisdiction, and no arbitrator shall have any authority to consider or decide any issue concerning the validity or enforceability of such prohibition. Any issues concerning arbitrability of a particular issue or claim under this Arbitration Agreement (except for those concerning the validity or enforceability of the prohibition against class, collective, representative, or multi-plaintiff action arbitration and/or applicability of the FAA) shall be resolved by the arbitrator, not a court.

The Arbitrator shall have the authority to consider and rule on dispositive motions, such as motions to dismiss, or motions for summary judgment, in accordance with the standards and burdens generally applicable to such motions in federal district court, except that the Arbitrator may establish appropriate and less formal procedures for such motions at the Arbitrator's discretion consistent with the expedited nature of arbitration proceedings. The Arbitrator will allow the parties to conduct adequate discovery including, but not limited to, issuing subpoenas to compel the attendance of witnesses at the arbitration hearing; serving written discovery; conducting depositions; and compelling the production of documents during discovery.

Covered Claims covered under this Arbitration Agreement include, but are not limited to: breach of contract, any claims challenging the independent contractor status of DISTRIBUTOR, claims alleging that DISTRIBUTOR was misclassified as an independent other claims premised contractor. anv upon DISTRIBUTOR's alleged status as anything other than an independent contractor, tort claims, discrimination claims, retaliation claims, and claims for alleged unpaid compensation, civil penalties, or statutory penalties under either federal or state law.

This Arbitration Agreement does not cover claims relating to whistleblowers and/or unlawful retaliation arising under the Sarbanes-Oxley Act or disputes involving any ERISA-based benefit plans that provide for arbitration. This Arbitration Agreement also does not preclude either DISTRIBUTOR or COMPANY from seeking provisional remedies such as temporary restraining orders or preliminary injunctions in accordance with applicable law. A party's seeking or obtaining such provisional remedies shall not be considered a waiver of that party's right to arbitration under this Arbitration Agreement.

Nothing in this Arbitration Agreement is intended to affect or limit DISTRIBUTOR's right to file an administrative charge or otherwise seek relief from any administrative or federal or state government agencies (although if DISTRIBUTOR chooses to pursue a claim following the exhaustion of such administrative remedies, that claim would be subject to the provisions of this Arbitration Agreement).

The parties agree that arbitration proceedings are to be treated as confidential, and that the parties will act to protect the confidentiality of the proceedings. The parties agree that neither they nor their counsel will reveal or disclose the substance of the arbitration proceedings, or the result, except as required by subpoena, court order, or other legal process. If disclosure is compelled of one party by subpoena, court order or other legal process, or as otherwise required by law, the party agrees to notify the other party as soon as notice of such process is received and before disclosure takes place. The parties may, however, disclose such information to their legal representatives, accountants or tax advisors as necessary so long as they agree to maintain such information in strict confidence. COMPANY may also disclose such information to individuals in affiliated companies for legal reporting purposes and other legitimate business reasons.

Any request for arbitration must be in writing and provided to the other party and to AAA by certified or registered mail, return receipt requested, within the time period provided for by the statute(s) of limitations applicable to the claim(s) asserted. The request must set forth a statement of the nature of the dispute, including the alleged act or omission at issue; the names of all persons involved in the dispute; the amount in controversy, if any; and the remedy sought.

DISTRIBUTOR acknowledges that this is an important document that affects its legal rights and that the DISTRIBUTOR has been given the opportunity to discuss this Arbitration Agreement with private legal counsel. If any provision of AAA's Rules or of this Arbitration Agreement are determined to be unlawful, invalid, or unenforceable, such provisions shall be enforced to the greatest extent permissible under the law, or, if necessary, severed, and all remaining terms and provisions shall continue in full force and effect. This Arbitration Agreement may be modified or terminated by COMPANY after thirty (30) days written notice to DISTRIBUTOR. Any modifications or terminations shall be prospective only and shall not apply to any claims or disputes that are pending in arbitration or that have been initiated by either party pursuant to the AAA Rules. The parties also agree that nothing herein is intended to, or does, affect or otherwise change the independent contractor relationship between them and that adequate and sufficient consideration has been provided for in this Arbitration Agreement, including but not limited to the additional monetary consideration and various other amendments to the Distributor Agreement as set forth in the Amendment executed concurrently herewith, and each party's promise to resolve their claims by arbitration. Finally, this Arbitration Agreement is the complete agreement of the parties on the subject of arbitration of disputes and supersedes any prior or contemporaneous oral or written agreement or understanding on the subject. Any agreement contrary to the foregoing must be in writing signed by the President of COMPANY.

This Arbitration Agreement shall be governed by the FAA and *Connecticut* law to the extent *Connecticut* law is not Inconsistent with the FAA.

DISTRIBUTOR acknowledges that it has received and read and specifically agrees to be bound by this Arbitration Agreement. DISTRIBUTOR understands that this Arbitration Agreement requires that disputes that involve matters subject to the Agreement be submitted to arbitration pursuant to the Arbitration Agreement rather than to a judge or jury in court and that such disputes must be brought on an individual basis only.

Date: 6/7/17

(Accepted)

DISTR	RIBUTOR	CON	IPANY	
s/Neal	Bissonnette	s/Mi	ke Brock	
By: N E	Veal Bissonnette	By:	Mike Brock	
Its: F	President	Its:	Vice President	
DISTR	RIBUTOR	COM	IPANY	
s/Neal Bissonnette		s/Mike Brock		
--------------------	---------------------	--------------	----------------	--
By:	Neal Bissonnette	By:	Mike Brock	
Its:	President	Its:	Vice President	

4 ≤ 2 -فتقر

71

EXHIBIT N — RECEIPT

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If CK Sales Co., LLC offers you a franchise, it must provide this disclosure document to you 14 calendar days before you sign a binding agreement with, or make payment to, the franchisor or an affiliate in connection with the proposed franchise sale.

If CK Sales Co., LLC does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal and state law may have occurred and should be reported to the Federal Trade Commission, Washington, DC 20580 and the state agency listed on Exhibit A.

The franchisor is **CK Sales Co., LLC**, located at 11 Adamian Drive, Auburn, Maine 04210. Its telephone number is (207) 783-9161.

Issuance date: April 1, 2017

The franchise sellers for this offering are Brian Devlin, Judd Price, Mike Brock and Jake Linthicum, 11 Adamian Drive, Auburn, Maine 04210.

CK Sales Co., LLC authorizes the respective state agencies identified in Exhibit A to receive service of process for it in the particular state.

I received a disclosure document dated *April 1, 2017* that Included the following Exhibits:

Exhibit A — State Regulators; Agents for Service

Exhibit B — Affiliated Companies

Exhibit C — Distributor Agreement.

The Distributor Agreement contains eleven (11) Exhibits:

- (i) Territory
- (ii) Products
- (iii) Authorized Products
- (iv) Bill of Sale
- (v) Owner
- (vi) Personal Guaranty
- (vii) Settlement Statement Authorization
- (viii) Distributor Choice of Insurance Coverages
- (ix) Distributor Choice of Delivery Location
- (x) Distributor Choice of Fee for Service
- (xi) Arbitration Agreement

Exhibit D — Territory Pro Forma

Exhibit E — Financing Documents

- (i) Secured Promissory Note and Personal Guaranty
- (ii) UCC-1

Exhibit F — Form of Purchase Agreement and General Release

Exhibit G — Form of Company Approval of Assignment and General Release

Assignment and General Release

Exhibit H — Form of Assignment/Assumption Agreement and Bill of Sale

- Exhibit I Current Independent Distributors
- Exhibit J Independent Distributor Terminations

Exhibit K — Previous Owner Information

Exhibit L — Financial Statements

Exhibit M — Form of Guarantee

Exhibit N — Receipts

	Neal	s/Neal
Date: <u>May 24, 2017</u>	Bissonnette	Bissonnette
(Do not leave blank)	Print Name	Signature

You may return the signed receipt either by signing, dating, and mailing it to CK Sales Co., LLC at 11 Adamian Drive, Auburn, Maine 04210, or by faxing a copy of the signed and dated receipt to Brian Devlin or Jake Linthicum at (207) 784-4634.

ATTACHMENT 2

75

DISTRIBUTOR AGREEMENT TABLE OF CONTENTS

ARTICL		PAGE
E		
I.	Warranty	1
II.	Definitions	2
III.	Relationship	4
IV.	Sale of Products	5
V.	Best Efforts/Distributor	5
VI.	Best Efforts/Company	6
VII.	Accounts/Trademarks Exempt	7
	From Agreement	
VIII.	Payment for Products	7
IX.	Equipment and Insurance	9
X.	Proprietary Services	9
XI.	Product Delivery	11
XII.	Product Code	11
XIII.	Advertising and Promotions	12
XIV.	Service Requirements	12
XV.	Transfer or Sale of Rights	12
XVI.	Independent Business	14
XVII.	Termination by Company	15
XVIII.	Dispute Resolution	16
XIX.	Trademarks, Trade Names, and	18
	Proprietary Materials	
XX.	Miscellaneous	20
Exhibit	Territory	24
Α		
Exhibit	Products	25
В		
Exhibit	Authorized Products	26
С		

Exhibit	Bill of Sale	27
D		
Exhibit	Owner	28
Ε		
Exhibit	Personal Guaranty	29
\mathbf{F}		
Exhibit	Settlement Statement	30
G	Authorization	
Exhibit	Distributor Choice of Insurance	32
Н	Coverages	
Exhibit I	Distributor Choice of Delivery	33
	Location	
Exhibit J	Distributor Choice of Fee for	34
	Service	
Exhibit	Arbitration Agreement	35
K		
Maryland Addendum (if applicable)		
Indiana Addendum (if applicable)39		

DISTRIBUTOR AGREEMENT

This Distributor Agreement ("Agreement") is made effective the <u>7th</u> day of <u>May, 2018</u>, by and between <u>CK</u> <u>SALES CO., LLC</u>, with its office and principal place of business at <u>11 Adamian Drive Auburn, ME 04210</u> ("COMPANY"), and <u>Blue Star Distributors Inc</u>, a *Connecticut* corporation, with its principal place of business at <u>78 Andrews Street Bristol, CT 06010</u> ("DISTRIBUTOR").

WITNESSETH:

COMPANY has developed or acquired a license or franchise for the use of formulae, recipes, trademarks and trade names through which it manufactures and/or markets bread, rolls, and other fresh-baked products, as defined herein; and Whereas, COMPANY, through the expenditure of time and effort and through its continuing research and marketing programs, has over the years developed a reputation for excellence in quality, value, and superior service which has created strong consumer recognition, approval, and demand for its products; and

Whereas, COMPANY has determined that the divestiture and division of its market area into distribution territories, and the sale of its products to independent franchise distributors for distribution and ultimate retail sale in said territories, will result in a superior distribution system; and

Whereas, DISTRIBUTOR is an independent contractor with the resources, expertise and capability to act as a franchise distributor of COMPANY's products in the Territory; and

Whereas, the parties desire to create a right in DISTRIBUTOR, under which DISTRIBUTOR will be authorized to sell certain defined products within a territory or territories and otherwise operate the distributorship business hereunder.

Now, therefore, in consideration of these premises, and of the covenants and conditions contained in this Agreement, and for other good and valuable consideration given and received, the receipt and the sufficiency of which are hereby specifically agreed to and acknowledged, the parties mutually agree as follows:

I. WARRANTY

1.1 The foregoing preambles are incorporated herein by reference and shall constitute warranties, representations, agreements and undertakings by the parties.

II. DEFINITIONS

- 2.1 *Outlets*: Shall mean all retail stores (except thrift stores) selling food to the general public and all restaurant, fast food, military and institutional accounts, in each case which allow direct store delivery (DSD) of Products and/or Authorized Products, except those exempt from this Agreement as set forth in Article VII, below.
- 2.2 **Products**: Shall mean fresh baked goods, as specifically described in **Exhibit B**, attached hereto and made a part hereof, which are currently produced and/or distributed by COMPANY, provided, however, that the Distribution Rights granted herein shall survive only so long as COMPANY and/or its affiliate(s) has/have the legal right to produce and sell such goods, whether by ownership, license, franchise, or otherwise, and COMPANY produces and/or sells such goods. Products shall not include products other than those specifically described in **Exhibit B** nor products intended to be sold as frozen or refrigerated.
- 2.3 Authorized Products: Shall mean fresh baked goods, as specifically described in Exhibit C, attached hereto and made a part hereof, provided, however, that DISTRIBUTOR does not acquire any proprietary or ownership rights to such Authorized Products, exclusive or non-exclusive. COMPANY may cease selling such Authorized Products to DISTRIBUTOR upon notice, with or without cause. COMPANY may also add Authorized Products to upon notice to DISTRIBUTOR. Exhibit C Authorized Products may include products with logos and/or labels, or without.

- 2.4 **Distribution Rights**: Except as expressly limited by this Agreement, Distribution Rights shall mean the right to sell and distribute Products to Outlets in the Territory, which right has been purchased by DISTRIBUTOR from COMPANY as evidenced by a Bill of Sale executed by the parties, which is attached hereto as **Exhibit D**, and made a part hereof.
- 2.5 *Territory*: Shall mean that geographic area as more specifically described in Exhibit A within which DISTRIBUTOR owns the Distribution Rights. By signing the Agreement, DISTRIBUTOR warrants and expressly represents that the Owner (as defined below) has personally driven the Territory description and confirms that the Territory description in Exhibit A is correct and reflects DISTRIBUTOR's understanding of the geographic of the Distribution Rights scope that DISTRIBUTOR is purchasing. DISTRIBUTOR hereby agrees to defend, indemnify, and hold harmless COMPANY and its parent, affiliates, and and their respective subsidiaries. directors. managers, shareholders, officers, employees, agents, and representatives from and against any and all claims, damages, lawsuits, judgments, losses, liabilities, penalties, fines, costs, and expenses (including reasonable attorneys' fees) incurred as result of any claim and/or regulatory or legal proceeding in which it is alleged that the Territory description in Exhibit A contains a mistake or is otherwise incorrect.
- **2.6** *Good Industry Practice*: Shall mean the standards that have developed and are generally accepted and followed in the baking industry, including, but not limited to, maintaining an adequate and fresh supply

of Products and Authorized Products in all Outlets in the Territory requesting service, actively soliciting all Outlets in the Territory not being serviced, properly rotating all Products and Authorized Products, promptly removing all stale Products and Authorized Products, maintaining proper service and delivery to all Outlets in the Territory requesting service in accordance with Outlet's requirements, maintaining all equipment in a sanitary condition and in good safe working order, and operating the distributorship business hereunder in compliance with all applicable federal, state and local laws, rules and regulations, including but not limited to, all motor vehicle, Department of Transportation, food, drug, health, bioterrorism, security, and sanitary laws and regulations. **DISTRIBUTOR specifically** acknowledges and agrees that such standards do not reflect control by COMPANY as to the specific details or manner and means of DISTRIBUTOR'S business, but reflect only COMPANY'S interests in the results achieved by DISTRIBUTOR, protecting the reputation of the brands, and protecting the business reputation of both **DISTRIBUTOR and COMPANY.**

- 2.7 *Stale*: Shall mean Products and Authorized Products removed from the market on the appropriate color code or designated pick up date as designated by COMPANY
- **2.8** *Out of Code*: Shall mean Products and Authorized Products left in the market beyond the appropriate color code or designated pick up date.
- **2.9** *Chain*: Shall mean a person or business entity that operates more than one Outlet and/or which makes centralized decisions regarding the purchase of

Products and/or Authorized Products for more than one Outlet.

2.10 *Owner*: Shall mean the individual holding a majority ownership in the stock of DISTRIBUTOR and listed on Exhibit E hereto and who is empowered to make binding decisions on behalf of DISTRIBUTOR related to modifications to the terms of this Such individual must personally Agreement. guarantee the conditions and obligations herein by executing the Personal Guaranty, attached hereto as Exhibit F and made a part hereof. As long as DISTRIBUTOR owns the Distribution Rights hereunder, Owner agrees to ensure DISTRIBUTOR is (i) properly established, organized and at all times in good standing with the appropriate state agency(s); (ii) in possession of all trading and other licenses, registrations, authorizations, consents, and permissions necessary or appropriate for the performance of DISTRIBUTOR's obligations as contemplated by this Agreement; and (iii) otherwise in compliance with appropriate state and federal law(s). Upon request, DISTRIBUTOR shall provide DISTRIBUTOR documentation showing that its corporate registration is in good standing and in compliance with appropriate state law(s).

III. RELATIONSHIP

3.1 *Extent and Duration*: COMPANY hereby recognizes DISTRIBUTOR's ownership of the Distribution Rights, said ownership to continue until:

(a) transferred, assigned or sold by DISTRIBUTOR or anyone acting on behalf of DISTRIBUTOR (this includes a sale by COMPANY for the account of DISTRIBUTOR in the event of a termination of this Agreement upon the terms defined below), or

(b) COMPANY, for legitimate business reasons unrelated to Section 3.1(c) immediately below, ceases to use distributors to distribute Products in the Territory. In such event, COMPANY will repurchase the Distribution Rights from DISTRIBUTOR at ten (10) times the average weekly sales volume of Products (not Authorized Products) in the Territory calculated over the six (6) month period preceding the repurchase, or

(c) either party exercises its right to terminate pursuant to Section 16.1. In such event, COMPANY will repurchase the Distribution Rights from DISTRIBUTOR at fair market value, which in no event shall exceed the purchase price paid by DISTRIBUTOR for the Distribution Rights.

3.2 *Nature of Rights*: The parties agree that the Distribution Rights sold to DISTRIBUTOR pursuant to the Bill of Sale can be exercised only pursuant to the terms of this Agreement and that any termination of this Agreement other than ln accordance with Section 3.1 (b) or (c) requires DISTRIBUTOR, or anyone acting on behalf of DISTRIBUTOR (Including a sale by COMPANY on behalf of DISTRIBUTOR as provided for in Section 17.4), to sell such Distribution Rights subject to the terms of this Agreement.

IV. SALE OF PRODUCTS AND AUTHORIZED PRODUCTS

4.1 Products and Authorized Products will be sold to DISTRIBUTOR at such terms and prices as established by COMPANY from time to time. Title

and risk of loss shall pass to DISTRIBUTOR upon delivery to DISTRIBUTOR as set forth in Section 11.1 below.

V. BEST EFFORTS/DISTRIBUTOR

5.1**Obligations of DISTRIBUTOR:** DISTRIBUTOR agrees and covenants to use DISTRIBUTOR's commercially reasonable best efforts to develop and maximize the sale of Products to Outlets within the Territory and service the Territory in accordance with Good Industry Practice as defined above, including servicing all Outlets in accordance with the Outlet's then-established service requirements. DISTRIBUTOR further agrees and covenants to use DISTRIBUTOR's commercially reasonable best efforts to maximize the sale of Authorized Products to those Outlets in the Territory for which such Authorized Products have been produced and to distribute such Authorized Products to such Outlets in accordance with Good Industry Practice as defined above, including servicing all Outlets in accordance with the Outlet's then-established service requirements. DISTRIBUTOR shall cooperate with COMPANY on its marketing and sales efforts and ensure its employee(s) maintain a clean and neat personal appearance consistent with the professional image customers and the public associate with and COMPANY, customer requirements. DISTRIBUTOR may not sell products in the Territory which are competitive with the Products or Authorized Products. DISTRIBUTOR may sell noncompetitive products, as long as this does not interfere with the sale of the Products or Authorized Products in the Territory. DISTRIBUTOR specifically acknowledges and agrees that the obligations herein do not reflect control by COMPANY as to the specific details or manner and means of DISTRIBUTOR'S business, but reflect only COMPANY's interests in the results achieved by DISTRIBUTOR, protecting the reputation of the brands, and protecting the business reputation of both DISTRIBUTOR and COMPANY. As set forth in Section 2.10, DISTRIBUTOR must be properly established, organized and at all times in good standing with the appropriate state agency(s) and in compliance with appropriate state law(s).

5.2 Alternate Distribution: Should DISTRIBUTOR believe a certain account or accounts is/are unprofitable to service, DISTRIBUTOR shall provide COMPANY written notice of such belief and the reasons therefor, including a detailed financial analysis. Within ten (10) business days thereafter, COMPANY representative(s) and DISTRIBUTOR shall meet to discuss DISTRIBUTOR's concerns. DISTRIBUTOR and COMPANY agree to work in good faith to explore opportunities to remedy the unprofitability such account(s), of and DISTRIBUTOR further agrees to give its best effort to remedy the unprofitability of such account(s), including the exercise of any recommendations made by COMPANY. After exhausting all such efforts, if the COMPANY agrees with DISTRIBUTOR's position, DISTRIBUTOR shall be relieved of its contractual obligation to service such account(s) for a period of time determined by COMPANY. During such period, COMPANY may make alternate distribution arrangements for such account(s) and DISTRIBUTOR shall not receive any credit for sales associated with such alternate distribution. At the

end of the period specified by COMPANY, COMPANY and DISTRIBUTOR shall meet again to determine future service of such account(s), including DISTRIBUTOR resuming service once such account(s) become profitable.

5.3 *Non-Compliance*: Failure to comply with Section 5.1 and/or any of the terms, conditions and obligations elsewhere in this Agreement shall be considered a material breach of this Agreement and shall be governed by the provisions of this Agreement dealing with termination.

VI. BEST EFFORTS/COMPANY

6.1 **Obligations of Company**: COMPANY shall use its commercially reasonable best efforts to manufacture and deliver to DISTRIBUTOR sufficient quantities of Products and Authorized Products to supply Outlets requesting service in the Territory, preserve and develop the quality and marketability of the Products, and cooperate with DISTRIBUTOR on DISTRIBUTOR's sales and marketing efforts. In order to assist DISTRIBUTOR with product sales to Chains, which may require COMPANY and/or its affiliate(s) involvement to obtain authorization to sell Products or Authorized Products, DISTRIBUTOR hereby designates COMPANY and/or its affiliate(s) and COMPANY and/or its affiliate(s) agree(s) to act as DISTRIBUTOR's limited agent to obtain such authorization. This limited agency designation is for the sole purpose of authorization discussions (Including space, position and pricing) with regard to Chain accounts. Nothing herein shall obligate COMPANY and/or its affiliate(s) to seek otherwise business authorization or pursue opportunities with any specific Chain or other account. Additionally, nothing herein shall prevent DISTRIBUTOR from having the right to deal directly with Chain accounts with regard to such terms, although DISTRIBUTOR acknowledges that (i) Chains may decide not to deal with either COMPANY or DISTRIBUTOR, and (ii) employees of Chains at the store and regional level may not have the authority enter into any agreements with DISTRIBUTOR.

6.2 Non-Compliance: Failure to comply with Section 6.1 shall be considered a material breach of this Agreement and shall be governed by Section 20.2 below (Company Breach). Repeated breaches by COMPANY which threaten to do substantial harm to DISTRIBUTOR'S business shall entitle DISTRIBUTOR to terminate this Agreement.

VII. ACCOUNTS AND/QR TRADEMARKS EXEMPT FROM DISTRIBUTORSHIP

- 7.1 Accounts Exempt From Agreement: COMPANY reserves the right to solicit and service drop delivery and other non-DSD accounts with Products and Authorized Products, in whole or in part, in DISTRIBUTOR's Territory, so long as it is done for legitimate business reasons and not for the purpose of undermining DISTRIBUTOR's business.
- **7.2** *Thrift Stores*: COMPANY reserves the right to continue to sell Products and Authorized Products through its retail thrift store operation, COMPANY-owned and/or independently owned.
- **7.3** *Other Trademarks*: The parties hereto stipulate that COMPANY and COMPANY affiliates produce a variety of products marketed under a variety of trademarks distributed through multiple channels of

distribution, Including warehouse distribution. This Agreement does not restrict any other distribution of products by COMPANY and COMPANY affiliates marketed under trademarks not listed on **Exhibit B** via warehouse distribution or otherwise.

VIII. PAYMENT FOR PRODUCTS

- 8.1 Settlement of Account: On or before Friday of each week, DISTRIBUTOR will remit to COMPANY a full settlement of all Products and Authorized Products sold to DISTRIBUTOR in the preceding week, in accordance with terms established by COMPANY from time to time. DISTRIBUTOR authorizes COMPANY to make credits and charges to its weekly settlement as set forth on the Settlement Statement Authorization, a sample of which is attached hereto as Exhibit G.
- 8.2 *Cash Sales*: DISTRIBUTOR is solely responsible for the collection of all cash sales in the Territory.
- 8.3 Non-Cash Sales: In cases where Products and/or Authorized Products are sold and distributed to Outlets which have been approved by COMPANY for credit and in its sole discretion, COMPANY will accept electronic data with corresponding proof of delivery, charge slips, Outlet-generated authorizations or other forms of payment authorizations as may be required by the Outlet or COMPANY (including but not limited to. authorizations signed by the Outlet's authorized representative, such as the store manager, or its lieu and designee) in of cash, credit DISTRIBUTOR's account for such sales, provided DISTRIBUTOR fully complies with COMPANY's credit policies as established from time to time. In the event DISTRIBUTOR fails to comply with

COMPANY's credit policy, fails to provide complete documentation as required by the Outlet or COMPANY, or falsifies any documentation or credit information, COMPANY shall be entitled to charge DISTRIBUTOR's account for any credit extended to DISTRIBUTOR resulting therefrom without limiting any other remedies available to COMPANY. DISTRIBUTOR is wholly responsible for collection of accounts receivable not authorized by COMPANY. COMPANY is obligated to credit DISTRIBUTOR only for such payments made by the Outlet.

- 8.4 *Chain Accounts*: COMPANY and/or its affiliate(s) reserves the right to continue carrying the accounts receivable for all Chain and other major accounts and to promptly credit DISTRIBUTOR for DISTRIBUTOR's sales to all such accounts.
- 8.5 Scan Based Trading: In the event a Chain account implements Scanned Based Trading (SBT), Pay By Scan (PBS) or accounting methodologies and/or technology, DISTRIBUTOR agrees to comply with policies and procedures as may be necessary to comply with such Chain accounting and/or technology requirements, including the provisions of Section 8.3 herein. In the event DISTRIBUTOR receives credit for products that are not accounted for by the Outlet's scanner data or an inventory of products in the Outlet ("Shrink"), COMPANY shall be entitled to charge DISTRIBUTOR's account to recovery any such credit that was wrongfully extended to DISTRIBUTOR.
- 8.6 Security Interest: To secure the prompt payment and timely performance of all indebtedness, obligations and liabilities of DISTRIBUTOR to COMPANY under this Agreement, whether now

existing or hereafter arising, DISTRIBUTOR hereby grants and conveys to COMPANY a presently existing and continuing security interest in the Distribution Rights, in this Agreement, in all Products and/or Authorized Products now or hereafter in DISTRIBUTOR's possession, in all accounts now or hereafter arising out of the sale by DISTRIBUTOR of Products and/or Authorized Products, and in all proceeds thereof. DISTRIBUTOR agrees that this security interest attaches immediately upon execution of this Agreement by DISTRIBUTOR and that COMPANY has all of the rights of a secured party under the applicable Uniform Commercial Code, as amended DISTRIBUTOR expressly from time to time. authorizes COMPANY to file and refile all appropriate Uniform Commercial Code financing statements necessary to perfect the security interests granted hereunder.

8.7 **Default**: Nothing herein shall be deemed to require COMPANY to fill an order of DISTRIBUTOR during the time when DISTRIBUTOR has failed to make any payment due to COMPANY in a timely fashion.

IX. EQUIPMENT AND INSURANCE

9.1 *Delivery Vehicle(s)*: DISTRIBUTOR is responsible for obtaining DISTRIBUTOR'S own delivery vehicle(s) and purchasing adequate insurance described in Section 9.2 below. thereon. asConsistent with industry good practice, DISTRIBUTOR agrees to keep the delivery vehicle(s) clean at all times and in a manner consistent with the professional image customers and the public associates with COMPANY, and to maintain the delivery vehicle(s) in such condition as to provide safe, prompt, and regular service to all customers. If legally required, DISTRIBUTOR agrees to have painted in a conspicuous manner on any delivery vehicle owned or leased by it to carry out the terms hereof: "Owned and Operated by (DISTRIBUTOR's name), An Independent Contractor", along with any other identifying information or registration number required by any state or federal motor carrier regulatory agency.

- 9.2 Insurance: DISTRIBUTOR shall maintain at all times throughout the duration of this Agreement insurance policies from insurance companies that maintain a financial strength rating of A- or better and are licensed to do business in the state(s) in which the Territory is located, and naming COMPANY as an additional insured party on all such insurance policies. DISTRIBUTOR may select either option as set forth in Exhibit H attached hereto and made a DISTRIBUTOR agrees to provide part hereof. COMPANY with written evidence of insurance and endorsement within three (3) days upon request by COMPANY reserves the right to COMPANY. change the insurance coverage requirements by providing DISTRIBUTOR with thirty (30) calendar days notice.
- **9.3** Other Insurance: In addition to the insurance coverage required under this Agreement, it is DISTRIBUTOR'S responsibility to carry and maintain any other insurance coverage DISTRIBUTOR may desire.

X. PROPRIETARY SERVICES

10.1 Services Provided: COMPANY shall make available, and DISTRIBUTOR shall use, certain

proprietary administrative services in order to assist DISTRIBUTOR in the conduct of DISTRIBUTOR's business for the following purposes: (i) collection of data; (ii) preparation of sales tickets: sales (iii) accumulation of sales histories; (iv) preparation of daily and weekly settlements; (v) preparation of "adds" to and "cuts" automated from DISTRIBUTOR's daily order of Products and Authorized Products; (vi) direct communication to COMPANY for the purpose of DISTRIBUTOR's ordering of product and receipt of daily load information; (vii) providing automated route book product information; (viii) providing automatic movement information; (ix) providing individual customer sales profiles; and (x) providing suggested orders for each customer. DISTRIBUTOR specifically acknowledges and agrees that the use of such proprietary administrative services is a necessary component of transacting business with Outlets and maximizing sales and does not reflect control by COMPANY of the specific details or manner and means of DISTRIBUTOR's business.

- **10.2** Administrative Fee: COMPANY shall charge, and DISTRIBUTOR shall pay, a fair market, reasonable administrative fee for these services, which shall be established from time to time by COMPANY. By signing below, DISTRIBUTOR specifically authorizes COMPANY to collect the Administrative Fee on a weekly basis via the weekly settlement statement process.
- **10.3** *Confidentiality*: DISTRIBUTOR agrees to maintain and hold in confidence any and all information obtained from or derived about

COMPANY's proprietary administrative services pursuant to the provisions of Section 20.7.

- **10.4** *No Proprietary Rights*: DISTRIBUTOR acknowledges that DISTRIBUTOR cannot and shall not acquire any proprietary rights in COMPANY's proprietary administrative services and no rights shall accrue to DISTRIBUTOR by virtue of the use of the proprietary administrative services. The right to use these proprietary administrative services shall not be assigned by DISTRIBUTOR without the express written consent of COMPANY.
- 10.5 Termination or Amendment of Services: This provision shall remain in effect for as long as this Agreement is in effect. However, in the event this Agreement is terminated for any reason, so shall DISTRIBUTOR's right to use the proprietary administrative services. COMPANY may, in its sole discretion, alter, amend or terminate the proprietary administrative services herein made available by the giving of fourteen (14) calendar days' notice to DISTRIBUTOR.
- Truthful 10.6 Accurate **Reporting**: and DISTRIBUTOR shall at all times utilize the proprietary services in an honest and ethical manner. This specifically includes accurate and truthful reporting of transactions and related information so as to allow COMPANY to comply with applicable legal disclosure and reporting requirements. COMPANY the right to audit reserves DISTRIBUTOR's transactional records and inventory to ensure accurate and truthful reporting and COMPANY's compliance with applicable legal requirements.

XI. PRODUCT DELIVERY

11.1 COMPANY shall deliver Products and Authorized Products to DISTRIBUTOR at the location chosen by DISTRIBUTOR, as reflected on Exhibit I attached hereto and made a part hereof. DISTRIBUTOR will be charged a fee for warehouse use. DISTRIBUTOR may request to change the location upon thirty (30) days' notice to COMPANY, provided COMPANY is able to accommodate the requested change, such decision being COMPANY's sole discretion. If COMPANY is not able to accommodate the requested change, it may provide DISTRIBUTOR with other location options. By DISTRIBUTOR signing below. specifically authorizes COMPANY to collect the warehouse fee on a weekly basis via the weekly settlement statement process.

XII. PRODUCT CODE

- 12.1 Maintaining a fresh market is a fundamental tenet of the baking industry. Accordingly, Out of Code Products or Authorized Products left in the market is a material breach of this Agreement. Repeated violations will be grounds for termination of this Agreement. DISTRIBUTOR specifically acknowledges and agrees that removing Stale Products and Authorized Products from the market on the appropriate color code or designated pick up date is necessary to maintain the reputation of the brands and to protect the business reputation of both DISTRIBUTOR and COMPANY.
- **12.2** DISTRIBUTOR is not obligated to sell its Stale Products and Authorized Products back to COMPANY. However, to assist DISTRIBUTOR in

maintaining a fresh market, COMPANY will repurchase a certain percentage of DISTRIBUTOR's Stale Products or Authorized Products, in accordance with COMPANY's Stale allowance policy as established by COMPANY from time to time.

12.3 DISTRIBUTOR may not sell any Out of Code Products or Authorized Products, or those not in a saleable condition for distribution to the general public, but may otherwise sell such products to purchasers for non-human consumption. DISTRIBUTOR may also choose to donate such products to a charity that will not resell the products, so long as such products are in a saleable condition.

XIII. ADVERTISING AND PROMOTIONS

- 13.1 COMPANY will provide all COMPANY initiated advertising material at no cost to DISTRIBUTOR, including, but not limited to, point of sale material and COMPANY scheduled media advertising. Subject to COMPANY'S prior approval, which approval will not be unreasonably withheld, DISTRIBUTOR may use other advertising materials.
- 13.2 DISTRIBUTOR will adhere to all promotions and feature pricing with respect to the major and Chain accounts in the Territory. Such promotions and feature pricing are optional with respect to local accounts. When DISTRIBUTOR follows promotions or feature pricing, DISTRIBUTOR will receive a reduction in the purchase price accounting for such promotion or feature pricing, such that DISTRIBUTOR and COMPANY share the price allowance. With respect to local accounts, DISTRIBUTOR shall be required to provide

COMPANY with sufficient proof that the promotion or feature pricing was extended to the local accounts(s) in order to obtain a pricing adjustment for such sales. Upon agreement of the COMPANY, DISTRIBUTOR and COMPANY will each share in any resulting pricing allowance(s).

XIV. SERVICE REQUIREMENTS

- **14.1** DISTRIBUTOR is responsible for providing proper service of the Territory at all times.
- 14.2 If DISTRIBUTOR does not service the Territory, for any reason, COMPANY reserves the right to service the Territory, and DISTRIBUTOR agrees to pay a daily fee as indicated on **Exhibit J**, plus any operating expenses COMPANY incurs, including but not limited to fuel, truck rental and travel costs. COMPANY reserves the right to change its daily fee. Such temporary service by COMPANY does not relieve DISTRIBUTOR of the obligation imposed on it by the AGREEMENT nor act to cure any breach by DISTRIBUTOR.

XV. TRANSFER OR SALE OF RIGHTS

15.1 Conditions: The Distribution Rights are owned by the DISTRIBUTOR and may be sold or transferred in whole or in part by DISTRIBUTOR, or anyone acting on behalf of DISTRIBUTOR, subject to the prior written approval of COMPANY, which approval shall not be unreasonably withheld. Any sale or transfer of the Distribution Rights must be bona fide, for legitimate business purposes, and compliant with all applicable laws. Additionally, the prospective purchaser or transferee must be fully qualified to meet all of the obligations under this Agreement, including but not

limited to, conducting the business as a corporation pursuant to the conditions set forth in Section 2.10. COMPANY's right of approval in this Section shall expire if not exercised within thirty (30) days after the later of (a) receipt by COMPANY of written notice from DISTRIBUTOR of its intent to sell or transfer to a named bona fide purchaser or transferee on terms and conditions fully set forth in such notice, and (b) an evaluation by COMPANY of the proposed purchaser or transferee. The transfer, whether in one or a series of transactions, of a majority of the ownership or voting interests in the stock of DISTRIBUTOR, whether by operation of law or otherwise, shall constitute a sale, subject to the prior written approval of COMPANY for purposes of this Article. DISTRIBUTOR shall execute and deliver to COMPANY and its affiliates a release of all its interests and claims to and in such Distribution Rights and all of its interests under or arising out of this Agreement, together with a general release of claims against COMPANY and its affiliates, in the event of any transfer or sale of the Distribution Rights, COMPANY will concurrently execute a general release of claims as to DISTRIBUTOR, except for any claim for monies due and owing COMPANY. Except as specifically set forth in Article III hereof, COMPANY has no obligation to repurchase the Distribution Rights.

15.2 Settlement of Account: No transfer or sale of DISTRIBUTOR'S rights under this Agreement is to be made unless and until DISTRIBUTOR has settled with COMPANY all outstanding accounts and settled all other outstanding liens and debts related to the distributorship; provided, however, that COMPANY may waive this requirement depending on the

circumstances of the particular transaction, including if it is satisfied that under the terms of the sale, the purchaser assumes all such liabilities and is financially capable of such assumption.

- 15.3 Transfer Fee: In the event of a sale by DISTRIBUTOR, or by anyone other than COMPANY acting on behalf of DISTRIBUTOR of DISTRIBUTOR's Distribution Rights to a third party, DISTRIBUTOR shall be solely responsible for ensuring the purchaser is fully qualified to operate the business hereunder, and shall pay a transfer fee to COMPANY in an amount equal to two percent (2%) of such gross sales price, but not less than \$2,000, in consideration of the administrative activities undertaken by COMPANY in connection with such sale. In the event of a sale of the Distribution Rights to COMPANY, or bv COMPANY on behalf of DISTRIBUTOR, DISTRIBUTOR shall pay a transfer fee to COMPANY in an amount equal to five percent (5%)of such gross sales price, but not less than \$3,000, in consideration of the administrative activities undertaken by COMPANY in connection with such sale.
- **15.4** *Taxes* and *Reporting Requirements*: DISTRIBUTOR shall be solely responsible for all taxes and transactional reporting requirements, including but not limited to the nature of the conveyance, related to the transfer or sale of the Distribution Rights.

XVI. INDEPENDENT BUSINESS

16.1 *Essential Term*: The status of DISTRIBUTOR pursuant to this Agreement is that of independent contractor for all purposes and the parties hereby

signify their express intention to this effect. DISTRIBUTOR shall not be controlled bv COMPANY as to the specific details or manner and means of DISTRIBUTOR's business, it being understood that the Interests of COMPANY are the results achieved by DISTRIBUTOR, protecting the reputation of the brands, and protecting the business reputation of both DISTRIBUTOR and COMPANY. DISTRIBUTOR'S business is separate and apart from that of COMPANY and it is of the essence of this Agreement that DISTRIBUTOR is an independent business. Any final determination that DISTRIBUTOR is not an independent contractor shall entitle either party to cancel this Agreement, such determination being contrary to the parties' express intention herein to create an independent contractor relationship. Neither DISTRIBUTOR nor any of DISTRIBUTOR'S employees, agents, or servants shall be considered or deemed in any way to be employees, agents or servants of COMPANY and neither party has the right or power, express or implied, to do any act or thing that would bind the other, except as herein specifically provided. DISTRIBUTOR shall not represent or do anything to suggest to its employees, agents, servants that they are employed by or work for COMPANY. The parties do not intend to act as joint employers, parent/subsidiary, joint venturers, or any other legal capacity other than separate and distinct businesses acting pursuant to the terms of this Agreement. There is no fiduciary relationship between the parties. Furthermore, none of the benefits provided by COMPANY to its employees are available from DISTRIBUTOR COMPANY to or to DISTRIBUTOR's employees, agents, or servants.

the DISTRIBUTOR and/or In event DISTRIBUTOR's employees, agents or servants hereafter become eligible to participate in any such benefits, DISTRIBUTOR, on behalf of DISTRIBUTOR and DISTRIBUTOR's employees, agents and servants, hereby waives any right to participate in such benefits. Such waiver is not dependent upon DISTRIBUTOR's status as an independent contractor. DISTRIBUTOR will be solely and entirely responsible for DISTRIBUTOR's acts and for the acts of DISTRIBUTOR's employees, agents, and servants during the performance of this Agreement, and will save and hold COMPANY harmless from any and all damages which may arise therefrom, including attorneys' fees.

- 16.2 Non-Personal Service: This Agreement does not require that DISTRIBUTOR's obligations hereunder be conducted personally by Owner or by specific individual in DISTRIBUTOR's anv DISTRIBUTOR shall be free to organization. engage such persons as DISTRIBUTOR deems assist appropriate to in discharging DISTRIBUTOR's responsibilities hereunder. Any breach of this Agreement by any person engaged by DISTRIBUTOR shall be deemed to be a breach by DISTRIBUTOR.
- **16.3** *Professional Services*: DISTRIBUTOR may engage any legal and/or accounting professional services it deems necessary for purposes of legal compliance, meeting its obligations under this Agreement, and otherwise.

XVII. TERMINATION BY COMPANY

17.1 *Performance*: Except as set forth in Sections 3.1 (b) and (c) and 16.1 above, or this Article, COMPANY

shall not terminate or cancel this Agreement, provided DISTRIBUTOR faithfully carries out the terms hereof. In the event DISTRIBUTOR fails to perform DISTRIBUTOR's obligations under this Agreement, COMPANY may terminate this Agreement as set forth below.

- 17.2 Non-Curable Breach: COMPANY may terminate upon twenty-four (24) hours' written notice and DISTRIBUTOR shall have no right to cure if: (i) DISTRIBUTOR'S failure of performance involves criminal activity, threatens public or private health or safety, involves violent activity or violations of law, or threatens to do substantial harm to COMPANY's business, trademarks or reputation, including, but any action or not limited to. inaction on DISTRIBUTOR's part results that in DISTRIBUTOR's inability to service any Chain account; (ii) DISTRIBUTOR falls to maintain the insurance coverage requirements as set forth in Section 9.2 above and Exhibit H: or (iii), DISTRIBUTOR dissolves its corporate status or Owner fails to maintain a majority ownership in the stock of DISTRIBUTOR;
- 17.3 *Curable Breach*: In any event of any other failure of performance by DISTRIBUTOR not described in Section 17.2, COMPANY must give DISTRIBUTOR ten (10) business days written notice within which DISTRIBUTOR may cure DISTRIBUTOR's failure of performance. If DISTRIBUTOR does not cure such failure of performance within this ten (10) day period, COMPANY may thereafter terminate this Agreement and DISTRIBUTOR shall have no further right to cure. Furthermore, the parties agree that repeated violations, even if cured, constitute a

chronic failure of performance and threaten substantial harm to COMPANY's business, trademarks or reputation, and in such event COMPANY shall be entitled to terminate this Agreement immediately and DISTRIBUTOR shall have no further right to cure.

17.4 Actions Following Termination: If this Agreement is terminated under either Section 17.2 or 17.3, COMPANY, within the limits of its ability to do so, will operate the business for the account of DISTRIBUTOR, deducting its reasonable expenses in connection with the operation thereof, and sell DISTRIBUTOR's Distribution Rights to a qualified purchaser(s) at the best price which can reasonably be obtained after proper notice and advertisement. Such sale shall be for the account of the terminated DISTRIBUTOR, and the proceeds of such sale, after deducting therefrom anv monies owed bv DISTRIBUTOR to COMPANY, the amount of any outstanding liens, any other known liabilities of the distributorship and the reasonable costs incurred in effecting the sale, shall be turned over to DISTRIBUTOR in exchange for the release of DISTRIBUTOR's Distribution Rights and interests under this Agreement, together with a general release of claims as to COMPANY and its affiliates. COMPANY will concurrently execute a general release of claims as to DISTRIBUTOR, except for any claim for monies due and owing COMPANY.

XVIII. DISPUTE RESOLUTION

18.1 *Negotiation*: DISTRIBUTOR and COMPANY shall attempt in good faith and employ their best efforts to resolve and terminate any controversy, claim or dispute arising out of or relating to this Agreement

promptly by negotiations. Such negotiations shall take place between representatives of COMPANY and DISTRIBUTOR who have the authority to settle and resolve the dispute. Such negotiations shall begin upon written notice from one party to the other describing any dispute or claim which has not been resolved in the ordinary course of business and suggesting a location for a meeting between the parties to conduct such negotiations within ten (10) business days after delivery of such notice. Representatives of DISTRIBUTOR and COMPANY shall meet at a mutually acceptable time and place within ten (10) calendar days after delivery of such notice and thereafter as often as they reasonably deem necessary, to exchange relevant information and to attempt to resolve the dispute. Any such meeting may be attended by one or more representatives of each party. No such meeting shall be attended by an attorney representing either party unless such party shall have first given the other party at least three (3) calendar days' notice that it will be accompanied by an attorney at the next scheduled meeting, and at such meeting the other party may also be accompanied by an attorney.

18.2 *Mediation*: If the dispute has not been resolved within thirty (30) calendar days of the initial notice, or if the party receiving such notice has failed to meet within fifteen (15) calendar days, either party may initiate mediation by a request therefore in writing to the other party. Upon receipt of such notice, DISTRIBUTOR or COMPANY shall be obligated to engage in mediation. If the parties fail to agree within fifteen (15) calendar days of the date of such request for mediation on the selection of a mediator, then the Center for Public Resources shall appoint a mediator in accordance with its Mediation Procedure and the parties shall continue efforts through mediation to resolve the controversy and dispute between them until mediation is terminated by the occurrence of any of the following events:

- (i) A written resolution in settlement of the dispute is reached, or
- (ii) The mediator informs the parties in writing that further efforts would not be productive or useful, or
- (iii) The parties agree in writing that further efforts would not be productive, or
- (iv) Sixty (60) calendar days elapse from the commencement of the mediation without resolution.

Neither COMPANY nor DISTRIBUTOR may withdraw from mediation before such a termination of the process.

18.3 Mandatory and Binding Arbitration: In addition to and/or instead of utilizing the negotiation and mediation procedures described in this Article, either DISTRIBUTOR or COMPANY may request binding arbitration to resolve any dispute between them. Both DISTRIBUTOR and COMPANY agree that all claims, disputes, and controversies arising out of or in any manner relating to: (i) this Agreement or any other agreement executed in connection with this Agreement: (ii) the performance. interpretation, application or enforcement hereof, Including, but not limited to breach hereof and/or termination hereof, (iii) the services provided to COMPANY by DISTRIBUTOR or by DISTRIBUTOR to COMPANY, or (iv) any other dealings between DISTRIBUTOR and COMPANY, which has not been resolved pursuant to the negotiation and mediation provisions herein shall be submitted to binding arbitration in accordance with the terms and conditions set forth in the Arbitration Agreement attached hereto as **Exhibit K**, excepting only such claims, disputes, and controversies as specifically excluded therein.

- **18.4** *Confidentiality*: All negotiations and mediations pursuant to this Article of the Agreement are confidential and shall be treated as compromise and settlement negotiations for purposes of the Federal Rules of Evidence and any similar state rules of evidence.
- **18.5** *Tolling*: All applicable statutes of limitation and defenses based upon the passage of time shall be tolled during the negotiation and mediation procedures specified in this Article. All deadlines specified in such procedures may be extended by mutual agreement of the parties.

XIX. TRADEMARKS, TRADE NAMES, AND PROPRIETARY MATERIALS

19.1 *Permission for Use*: Subject to the terms and conditions of this Article and to the Distribution Rights granted herein, for the Duration of this Agreement, COMPANY authorizes DISTRIBUTOR to use, on a non-exclusive basis, the trade names and trademarks (collectively, "Trademarks"), trade dress, package designs, logos, artwork, advertising materials and other works protected by copyright, and other intellectual property of or associated with the Products (collectively, including the Trademarks, the "Proprietary Materials") solely in connection with DISTRIBUTOR's advertising, promoting,

marketing, sale, and distribution of Products in the Territory. DISTRIBUTOR may not assign or transfer this authorization except in the context of a sale or transfer of all or a portion of the Distribution Rights as provided in this Agreement. DISTRIBUTOR will not use any of the Proprietary Materials in any manner different from how they are used generally by COMPANY.

19.2 Conditions of Use: DISTRIBUTOR shall make no use of the Trademarks or Proprietary Materials, nor engage in any program or activity which makes use of or contains any reference to COMPANY, its products, Trademarks or Proprietary Materials, other than as specifically provided herein, except with the prior written consent of COMPANY. Should DISTRIBUTOR advertise other services or products on its delivery vehicle, such advertising shall not dilute or negatively affect the reputation and goodwill of COMPANY, the Trademarks or the Proprietary Materials. DISTRIBUTOR will not (i) do anything that would or might be inconsistent with, impair, or conflict with ownership of the Proprietary Materials by COMPANY or, if licensed, franchised, or otherwise, the owner thereof; (ii) use any of the Proprietary Materials in any manner likely to (a) deceive or mislead the public, (b) endanger the validity or enforceability of any of the Proprietary Materials, or (c) damage or impair the reputation or value of COMPANY, any of the Proprietary Materials, or any Product; (iii) attack, contest, oppose, or interfere with the right, title, or interest in or to any Proprietary Materials of COMPANY or the owner thereof in any jurisdiction; or (iv) adopt or use any work confusingly or substantially similar to any of the Proprietary Materials. DISTRIBUTOR will
not use any of the Trademarks or Proprietary Materials in either its legal or fictitious trade name without the prior written consent of COMPANY. DISTRIBUTOR will not use any of the Trademarks, Proprietary Materials, or any variation thereof as a business and/or corporate name or d/b/a of DISTRIBUTOR, an Internet domain name or a mnemonic, cipher, or "vanity" telephone number without COMPANY's prior written approval.

- **19.3** *Monitoring and Cooperation*: To insure the uniform quality of the goods in connection with which the Proprietary Materials are used, COMPANY will have the right to require DISTRIBUTOR to Immediately discontinue any non-approved use of any Proprietary Materials or any other activity that COMPANY in its sole judgment considers likely to impair the value or validity of the Proprietary DISTRIBUTOR will immediately Materials. observe and require that other persons under its control observe any and all instructions of the COMPANY limiting or halting the use of the Proprietary Materials. DISTRIBUTOR will notify COMPANY, in writing, as soon as DISTRIBUTOR becomes aware of any infringement, imitation, impairment, counterfeiting, or other unlawful or improper use of any of the Proprietary Materials or any of the Products. DISTRIBUTOR will cooperate with and provide testimony and other evidence to COMPANY in and in connection with any application for, registration and protection of, or enforcement of the Proprietary Materials in the Territory as and when requested by COMPANY.
- **19.4** *Acknowledgment*: DISTRIBUTOR acknowledges and agrees that (i) all the rights concerning the

Proprietary Materials are owned by COMPANY or the owner thereof, (ii) DISTRIBUTOR does not have, and will not as a result of this Agreement or performance hereunder acquire, any ownership or legal rights in any of the Proprietary Materials used in connection with the Products, and (iii) DISTRIBUTOR's sole rights with respect to the Proprietary Materials are set forth in this Article.

19.5 *Return Upon Termination*: Upon termination of this Agreement, DISTRIBUTOR shall cease use of COMPANY's Proprietary Materials and shall deliver to COMPANY any advertising, promotional or merchandising material.

XX. MISCELLANEOUS

- 20.1 Notice: Any notice required or permitted under this Agreement shall be deemed properly given and received on the date received if hand delivered; on the next business day after deposit with a nationally recognized overnight courier service for overnight delivery; or three (3) days after deposited in the mails, return receipt requested, registered or certified mail postage pre-paid, addressed to the parties at the addresses first referenced above, or at such other address as either party may designate by written notice given pursuant to this section.
- **20.2** *Company Breach*: If DISTRIBUTOR maintains that COMPANY is in breach of this Agreement, DISTRIBUTOR should notify COMPANY in writing, by certified mail, return receipt requested. Such written notice must include the specific section of this Agreement DISTRIBUTOR maintains has been breached and sufficient facts to provide reasonable notice to COMPANY of the action or failure to act which DISTRIBUTOR maintains is a

breach of this Agreement. Upon receipt of the DISTRIBUTOR's notice of breach, COMPANY shall have a reasonable period of time to investigate and cure any breach. Any failure to comply with this provision shall not affect DISTRIBUTOR's right to submit a claim to mandatory and binding arbitration.

- 20.3 *Survival*: This agreement shall be binding upon the heirs, personal representatives, successors or assigns of the parties to this Agreement. Notwithstanding any termination of this Agreement, all provisions hereof that, by their terms or reasonable interpretation thereof, set forth obligations that extend beyond the termination of this Agreement, shall survive and remain in full force and effect.
- **20.4** *Incorporation of Bill of Sale*: This Agreement is subject to and affected by a Bill of Sale simultaneously executed by the parties, such Bill of Sale being fully incorporated in this Agreement by reference.
- 20.5 Entire Agreement: This Agreement, together with all Exhibits incorporated herein by reference, as amended from time to time in accordance with the Agreement, sets forth the entire agreement between the parties and supersedes all prior agreements, discussions, negotiations, understandings, representations, conditions, warranties and covenants between them with respect to this subject matter. Unless set forth in this Agreement, no party shall be liable for any representation made to any other. Except as otherwise expressly authorized herein, this Agreement may be amended or modified only by a writing signed by all parties.

- 20.6 Indemnification: DISTRIBUTOR agrees to indemnify and hold harmless COMPANY from and against any and all claims, actions, liabilities, expenses, losses or demands, including reasonable attorneys' fees, growing out of or based upon this Agreement acts or omissions or any of DISTRIBUTOR arising in the normal course of the conduct of business by DISTRIBUTOR pursuant to this Agreement. DISTRIBUTOR shall also indemnify COMPANY against all liability and loss in connection with, and shall assume full responsibility for, payment of all federal, state and local taxes and/or contributions imposed or required under unemployment insurance, workers' compensation insurance, and income tax laws with respect to DISTRIBUTOR or DISTRIBUTOR's employees engaged in performance of this Agreement.
- 20.7 Trade Secrets/Confidential Business Information: DISTRIBUTOR recognizes and acknowledges that COMPANY and its affiliated entities have, through the expenditure of substantial time, effort and money, developed and acquired certain trade secrets and confidential business information ("Confidential Information") which are of great value to COMPANY. DISTRIBUTOR further acknowledges and understands that through the distributor relationship with COMPANY, DISTRIBUTOR will have access to Confidential Information of COMPANY and its affiliated entities. Accordingly, except as and to the extent required by law, DISTRIBUTOR covenants that DISTRIBUTOR will make no disclosure, or any use, in any manner whatsoever, of any of the Confidential Information of COMPANY and/or any of its affiliated entities while this Agreement is in effect and thereafter, except as

specifically authorized by COMPANY. Upon termination of this Agreement, DISTRIBUTOR shall return to COMPANY all embodiments of such Confidential Information that are in DISTRIBUTOR's possession, custody or control. As used herein the term Confidential Information includes (a) sales data, financial statements and information, marketing arrangements and plans, trade secrets, pricing information and strategies, business development plans, and any other business methods, processes and techniques of COMPANY and/or its related entities to the extent all of which are not generally known to the trade or industry and/or (b) confidential or proprietary information or trade secrets of third parties, including retail customers, with which COMPANY and/or its related entities conducts business and which is supplied to DISTRIBUTOR for purposes of this Agreement. Notwithstanding the foregoing, Confidential Information shall not include information that: (i) is or becomes generally known to the public not as a result of a disclosure by DISTRIBUTOR; (ii) is received by DISTRIBUTOR in good faith and without restriction from a third party having the right to make such disclosure; and/or (iii) is known to DISTRIBUTOR at the time of COMPANY's thereof DISTRIBUTOR disclosure to or DISTRIBUTOR's becoming aware thereof in the course of and as a result of DISTRIBUTOR's business relationship with COMPANY. DISTRIBUTOR's obligations in this paragraph shall be in addition to, and not in lieu of, any and all obligations imposed upon DISTRIBUTOR by applicable law.

- 20.8 Injunctive Relief: DISTRIBUTOR agrees and acknowledges that a violation of Section 20.7 will cause irreparable harm to COMPANY and that there will be no adequate remedy at law to redress the resulting harm that will be suffered by COMPANY. Therefore, DISTRIBUTOR further agrees that in the event of any violation or threatened violation of this Section, COMPANY shall be entitled as a matter of course to an immediate injunction out of any court of competent jurisdiction restraining such violation or threatened violation by DISTRIBUTOR, such right to be cumulative and in addition to whatever other remedies, at law or in equity, that COMPANY may have.
- 20.9 Savings Provision/Non-Waiver: Should any portion, word, clause, sentence or paragraph of this Agreement be declared void or unenforceable, including the Arbitration Agreement attached hereto, such portions shall be modified or deleted in such a manner as to make this Agreement as modified legal and enforceable to the fullest extent permitted under applicable law. Except as expressly provided herein, no waiver by either party of any default in the performance of any part of this Agreement by the other party shall be deemed a waiver of any other default hereunder.
- **20.10** Legal Counsel: DISTRIBUTOR acknowledges that it has been given a reasonable opportunity to discuss this Agreement with an attorney of its choosing; that it has carefully read and fully understands the provisions of this Agreement, and that it is entering into this Agreement knowingly, voluntarily, and with the intention of being legally bound by all of the terms of this Agreement.

- **20.11** *Governing Law*: This Agreement and the construction thereof shall be governed by the laws of the State of *Connecticut*, without regard to the conflict-of-law rules.
- 20.12 *Revocation*: This Agreement, including the Arbitration Agreement attached hereto, may be revoked by DISTRIBUTOR within seven (7) calendar days after DISTRIBUTOR executes it ("Revocation Period") by sending written notice of revocation to *Peter Roy*, and the Agreement will not become effective or enforceable until this Revocation Period has expired, without revocation.
- 20.13 Right to Opt-Out of Mandatory and Binding Arbitration: DISTRIBUTOR may elect to opt-out arbitration procedures described in of the Article 18.3, above, and Exhibit K, attached hereto, within seven (7) calendar days after DISTRIBUTOR executes this Agreement ("Opt-Out Period") by sending written notice of such intent via hand delivery, email or fax to *Peter Roy*. Neither Article 18.3 nor Exhibit K will become effective or enforceable until this Opt-Out Period has expired and DISTRIBUTOR has not elected to opt-out of the arbitration procedures described in Article 18.3, above, and Exhibit K, attached hereto. Should DISTRIBUTOR choose to opt-out of the arbitration procedures described in Article 18,3, above, and Exhibit K, attached hereto, then neither Article 18.3 nor Exhibit K shall be part of this Agreement, neither DISTRIBUTOR nor COMPANY shall be required to comply with any of their provisions, and neither DISTRIBUTOR nor COMPANY will be required to submit any claims, disputes, and controversies between them to binding arbitration.

IN WITNESS WHEREOF, COMPANY and DISTRIBUTOR have executed this Agreement this 25th day of April, 2018, and each signatory hereto represents that he or she has the authority to execute this Agreement and legally bind the respective party.

DISTRIBUTOR

COMPANY

s/Tyler	Wojnarowski
By:	Tyler Wojnarowski
Its:	President

s/Judd E. Price

By: Judd E. Price Its: Market VP

WITNESSES: where Doriel

ATTESTI

EXHIBIT "A" — TERRITORY

Territory begins at intersection of Mrytle Street and Corbin Ave. Go south on Corbin Ave to Route 372. Turn right on Route 372 to Route 10. Turn left on Route 10 to West Main Street. Turn right on West Main Street to West Street. Turn right on West Street to Spring Street. Turn right on Spring Street to Route 10 working both sides throughout.

Territory begins at intersection of West Main Street and Route 10. Go south on Route 10 to King Road working both sides throughout.

Territory begins at intersection of Meriden Ave and Route 322. Go west on Route 322 to Shelton Ave working both sides throughout.

Acknowledged and Agreed:s/Tyler Wojnarowskis/Judd E. PriceDISTRIBUTORCOMPANY

EXHIBIT "B" — PRODUCTS

Except as expressly limited by the Distributor Agreement, DISTRIBUTOR shall have the exclusive right to sell and distribute the following Products in the Territory specifically described in **Exhibit A**.

- 1. Nature's Own
- 2. Country Kitchen
- 3. Barowsky's
- 4. TastyKake
- 5. Wonder Bread
- 6. Home Pride
- 7. Cobblestone Bread Company
- 8. Old Fashion Hearth

EXHIBIT "C" — AUTHORIZED PRODUCTS

Except as expressly limited by the Distributor Agreement, DISTRIBUTOR is authorized to distribute the following products. Authorized Products may include products with logos and/or labels, or without. DISTRIBUTOR does not acquire any proprietary or ownership rights to such Authorized Products, exclusive or non-exclusive, and COMPANY may cease selling such Authorized Products to DISTRIBUTOR upon notice. COMPANY may also add Authorized Products to this Exhibit upon notice to DISTRIBUTOR.

- 1. Best Yet
- 2. C&S
- 3. Clear Bag
- 4. HANNAFORD
- 5. Middle East
- 6. Shurfine
- 7. Sunbeam
- 8. SuperBuy
- 9. Great Value
- 10. Dave's Killer Bread Products

EXHIBIT "D" — BILL OF SALE

CK SALES CO., LLC, ("COMPANY"), subject to the consideration set forth below, effective the <u>7th</u> day of <u>May, 2018</u>, hereby conveys, sells, transfers and delivers to <u>Blue Star Distributors Inc</u>, ("DISTRIBUTOR"), the Distribution Rights specifically described in the Distributor Agreement executed simultaneously with this Bill of Sale. Both the Distributor Agreement and all appendices (Exhibits) thereto are specifically incorporated into and made a part of this Bill of Sale.

The terms of this sale are as follows:

Purchase price shall be *\$61,980,00*. DISTRIBUTOR shall own all Distribution Rights in the Territory consistent with the terms and conditions contained in the Distributor Agreement.

COMPANY has lawful title to the Distribution Rights in the Territory free from all encumbrances. COMPANY warrants and will defend the rights conveyed herein against the lawful claims and demands of all persons.

IN WITNESS WHEREOF, COMPANY and DISTRIBUTOR execute this Bill of Sale this <u>25th</u> day of <u>April, 2018</u>.

DIST	RIBUTOR	COMPANY				
s/Tyle	r Wojnarowski	s/Judd E. Price				
By:	Tyler Wojnarowski	By: J	udd E. Price			
Its:	President	Its: M	Iarket VP			

WITNESSES 10%

ATTEST

EXHIBIT "E" – OWNER

Owner's Name:	Tyler J. Wojnarowski
Address:	78 Andrews Street
City, State, ZIP	Bristol, CT 06010
TELEPHONE:	860.985.5989
Email Address:	two jo 65@sbcglobal.net

EXHIBIT "F" — PERSONAL GUARANTY

In consideration of the foregoing premises, covenants and conditions, and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, Tyler J. Wojnarowski, an individual ("GUARANTOR"), does hereby irrevocably and unconditionally guarantee to COMPANY, and become surety to COMPANY, for the performance and compliance with the terms, conditions and obligations of this Distributor Agreement entered into by and between Blue Star Distributors Inc (corporation Name), a ("DISTRIBUTOR"), Connecticut corporation and COMPANY. If any term, condition or obligation of the Distributor Agreement is not complied with, performed, or paid by DISTRIBUTOR as required therein, including any terms, conditions or obligations due following **GUARANTOR** termination thereof. will, upon COMPANY's demand, immediately ensure the timely and complete performance of DISTRIBUTOR of each and every obligation and duty imposed on it by the Distributor Agreement, and/or pay any amounts due and owing due to DISTRIBUTOR's breach of the Distributor Agreement or as required by the terms and conditions thereof. GUARANTOR shall also pay to COMPANY upon demand all costs and expenses, including but not limited to reasonable attorney fees, which may be incurred by COMPANY in the enforcement and/or collection of monies due hereunder. GUARANTOR is fully responsible for ensuring DISTRIBUTOR is properly established, organized and at all times in good standing with the appropriate state agency(s), and in compliance with appropriate state law(s) in accordance with Section 2.10 Agreement. of the Distributor GUARANTOR agrees and acknowledges he/she is subject

to the Arbitration Agreement attached hereto as **Exhibit K**.

s/Tyler Wojnarowski 4-25-18

GUARANTOR Signature Date

Tyler Wojnarowski GUARANTOR Name (Print)

WITNESSES:

Just

EXHIBIT "G" — SETTLEMENT STATEMENT AUTHORIZATION

I understand that on a weekly basis I will receive a settlement statement from COMPANY (in the format of the sample shown below). I also understand that this statement reflects an ongoing accounting between COMPANY and me (and the business entity pursuant to which my distributorship is operated), for product purchases, adjustment to product purchases, various credits, and business expenses.

I (personally and on behalf of my business entity) hereby authorize COMPANY to make the adjustments noted on the settlement statement, both charges and credits, on a weekly basis. I understand that my portion of customer price allowances is included in the standard discount.

I (personally and on behalf of my business entity) also authorize COMPANY to collect from me (and/or the business entity pursuant to which my distributorship is operated), via the weekly settlement statement, business expenses such as shown on the below sample, on a weekly basis. I understand that the amount for certain business expenses DISTRIBUTOR may incur, as applicable, including, for example, territory payment, truck lease, health/disability, life insurance, truck/business insurance, administrative fee and warehouse fee (as applicable) will remain the same each week and authorize COMPANY to deduct charges for these expenses from my settlement each week in exact amount show below.

I also understand that certain other business expenses, such as stale adjustment, truck rental, truck repair, truck tax, tag and license fees, sales and use tax, open route expense, supplies/equipment, apparel, temporary services fees, SBT shrink charges and miscellaneous expense listed below may vary weekly. By signing below, I (personally and on behalf of my business entity) also authorize deduction of any of these expenses, as applicable, each week.

Finally, I (personally and on behalf of my business entity) understand and agree that should I dispute any of the credits or charges on a weekly statement that I must bring such dispute to COMPANY's attention in writing within ten (10) calendar days of receipt of the statement with the disputed credit or charge. If I do not, I understand that my failure to do so indicates my agreement that all charges and credits on the statement are proper.

DISTRIBUTOR

WITNESS

s/Tyler Wojnarowski

De Sot

Total Product Charged Relay Adjustment Load Adjustment Damaged Short Code	Transferred Funds Paid To Distributor Funds Received From Distributor	Beginning Balance Owed Prior Weeks' Balances	Name: Distributor Number: Route: Warehouse: Date: Company Code:
<u>Accountability</u> 10,231.50 0.00 3.46- 0.00			Week # 08
0.00			(02/18/2018 to 02/24/2018)
Margin / Discounts		0.00	Master Distrib Name: Master Distrib Number: Master Route: VP of Sales: Director RSM: Branch:
	192.38 0.00	10,764.19 0.00	

Scan Based Trading Adj. (02/11/2018 to 02/17/2018) Point of Sale 5,090.72 Data Revaluation 107 Variance	Contest Earnings Other Earnings Authorized Charges AR Ticket Chargebacks	Product Transfers Cost Price Allwnce/Overrides Stales Standard Discount Additional
ij. (02/11/2018 5,090.72- 107.23-	2,523.05- 0.00	212.46 1,279.12- 1,387.38-

Tag, License, Fees Truck/Bus Insurance	Truck Tax,	Truck Repair	Truck Rental	Truck Lease	Purchase	Truck	Payment	Territory	Adjustment	Stale	Expenses:	Business	Balance	New Distributor	Shrink
														179.89-	226.91-
0.0	0.00	0.00	0.00	169.33		0.00		226.51		0.00					
0.00 Apparel	Supplies / Equipment	Open Route Expense	Adjustment Sales and Use Tax	FICA Tax-	Wk	FICA Tax-Current		Warehouse Fee		Administrative Fee				1,374.76-	
0.00	0.00	0.00	0.00	0.00		105.18		36.00		26.00					

 $1,\!548.65$ -

To be paid via Direct Deposit to bank account on record this Friday	ank account	Direct Deposit to k	paid via	To be]	
	138.89-				Weekly Distributor Balance
					02/24/2018):
	$4,\!578.12$ -	'18/2018 to	ets (02/	very Tick	Less Current Week SBT Delivery Tickets (02/18/2018 to
	5,531.71-	cy (02/17/2018):	nventor	Trading I	Less Prior Week Scan Based Trading Inventory (02/17/2018):
					Scan Based Trading Credits
	9,970.94				Total Distributor Balance
					Expense
					Business
563.02					Total
					Insurance
					Life
		Expense	0.00		License
	0.00	Miscellaneous		0.00	Health
		Charges			Insurance
	0.00	SBT Shrink		0.00	Disability
					Insurance
					Health
	0.00	Temp Syce Fee		0.00	

EXHIBIT "H" DISTRIBUTOR CHOICE OF INSURANCE COVERAGES

Option 1:

- (i) Automobile liability insurance, including hired auto and non-owned auto, on all vehicles used in the business, in such amounts as may reasonably be established by COMPANY from time to time, which as of the date of this Agreement will include combined single limit coverage of at least \$1,000,000; and collision and comprehensive loss coverage for the actual cash or replacement cost value of any and all delivery vehicles operated by DISTRIBUTOR that are financed or leased, with no more than a \$500 deductible. Such coverage shall be primary and not contributory to any other coverage available to COMPANY;
- (ii) Comprehensive general liability coverage, in such amounts as may reasonably be established by COMPANY from time to time, which as of the date of this Agreement will include an occurrence and aggregate limit of at least \$1,000,000. Such coverage shall be primary and not contributory to any other coverage available to COMPANY;
- (iii) Umbrella or excess liability coverage, in such amounts as may reasonably be established by COMPANY from time to time, which as of the date of this Agreement will include an occurrence and aggregate limit of at least \$1,000,000. Such coverage shall be primary and not contributory to any other coverage available to COMPANY;
- (iv) Workers compensation coverage for any and all employees of DISTRIBUTOR in compliance with all

requirements and laws of the state or states in which DISTRIBUTOR operates.

Option 2:

- (i) Automobile liability insurance, including hired auto and non-owned auto, on all vehicles used in the business, in such amounts as may reasonably be established by COMPANY from time to time, which as of the date of this Agreement will include combined single limit coverage of at least \$2,000,000; and collision and comprehensive loss coverage for the actual cash or replacement cost value of any and all delivery vehicles operated by DISTRIBUTOR that are financed or leased, with no more than a \$500 deductible. Such coverage shall be primary and not contributory to any other coverage available to COMPANY;
- (ii) Comprehensive general liability coverage, in such amounts as may reasonably be established by COMPANY from time to time, which as of the date of this Agreement will include an occurrence and aggregate limit coverage of at least \$2,000,000. Such coverage shall be primary and not contributory to any other coverage available to COMPANY;
- (iii) Workers compensation coverage for any and all employees of DISTRIBUTOR in compliance with all requirements and laws of the state or states in which DISTRIBUTOR operates.

EXHIBIT "I" DISTRIBUTOR CHOICE OF DELIVERY LOCATION

As set forth in Section 11.1, COMPANY will deliver Products and Authorized Products to DISTRIBUTOR at the location selected below by DISTRIBUTOR.

Please select choice below by initialing the appropriate selection:

TW	1.	Location #1 Waterbury, CT
	2.	Location #2 Bridgeport, CT
	3.	Location #3 Windsor, CT

131

EXHIBIT "J" DISTRIBUTOR CHOICE OF COMPANY FEE FOR SERVICE

As set forth in Section 14.2, if COMPANY services DISTRIBUTOR'S Territory, DISTRIBUTOR agrees to pay a daily fee to COMPANY as selected below by DISTRIBUTOR. By signing the Distributor Agreement, DISTRIBUTOR specifically authorizes COMPANY to collect the fee for service via the weekly settlement statement process,

Please select choice below by initialing the appropriate selection:

	Option 1:	<i>\$350</i> per day.
TW	Option 2:	12% of average daily net sales.

EXHIBIT "K" — ARBITRATION AGREEMENT

The parties agree that any claim, dispute, and/or controversy except as specifically excluded herein, that either DISTRIBUTOR (Including its owner or owners) may have against COMPANY (and/or its affiliated companies and its and/or their directors, officers, managers, employees, and agents and their successors and assigns) or that COMPANY may have against DISTRIBUTOR (or its owners, directors, officers, managers, employees, and agents), arising from, related to, or having any relationship or connection whatsoever Distributor with the Agreement between DISTRIBUTOR and COMPANY ("Agreement"), including the termination of the Agreement, services provided to COMPANY by DISTRIBUTOR, or any other association that DISTRIBUTOR may have with COMPANY ("Covered Claims") shall be submitted to and determined exclusively by binding arbitration under the Federal Arbitration Act (9 U.S.C. §§ 1, et seq.) ("FAA") in conformity with the Commercial Arbitration Rules of the American Arbitration Association ("AAA" or "AAA Rules"), or any successor rules, except as otherwise agreed to by the parties and/or specified herein. Such arbitration shall be conducted before a single arbitrator unless all parties to the arbitration agree otherwise in writing. Copies of AAA's Rules are available on AAA's website (www.adr.org).

COMPANY shall pay for all arbitration filing fees and costs that are customarily associated with AAA arbitration, subject to the Arbitrator's authority to award fees and costs to COMPANY as the prevailing party. Each party may be represented by legal counsel of their own choosing. Each party shall pay its own attorneys' fees, provided that an Arbitrator may award attorney's

fees and costs to the prevailing party under any applicable statute or written agreement to the same extent that attorney's fees and costs could be awarded in court. The arbitration shall be subject to the same burdens of proof and statutes of limitations as if the Covered Claims were being heard in court. The Arbitrator shall issue a written decision within forty-five (45) days of the later of: (1) the arbitration hearing; or (2) submission of the parties' postarbitration briefs. The Arbitrator's written decision shall include findings of fact and conclusions of law. The Arbitrator shall have the authority to award the same damages and other relief that would have been available in court pursuant to applicable law had the Covered Claim been brought on an individual basis in such forum, including attorneys' fees and costs. Subject to the parties' right to appeal, the decision of the arbitrator will be final and binding. The Arbitrator shall not have the authority to add to, amend, or modify, existing law. No arbitration award or decision will have any preclusive effect as to any issues or claims in any dispute, arbitration, or court proceeding where any party was not a named party in the arbitration.

All Covered Claims against COMPANY must be brought by DISTRIBUTOR on an individual basis only and not as a plaintiff or class member in any purported class, collective, representative, or multi-plaintiff action. DISTRIBUTOR further agrees that if it is within any such class, collective, representative, or multi-plaintiff action, it will take all steps necessary to opt-out of the action or refrain from opting in or joining, as the case may be, and DISTRIBUTOR expressly waives any right to recover any relief from any such class, collective, representative, or multi-plaintiff action. Similarly, all Covered Claims by COMPANY against DISTRIBUTOR may not be brought as a plaintiff or class member in any purported class, collective, representative, or multiplaintiff action. The parties understand that there is no right or authority for any Covered Claim to be heard or arbitrated on a multi-plaintiff, collective, or class action basis, as a private attorney general, or any other representative basis. The parties understand that there are no bench or jury trials and no class, collective, representative, or multi-plaintiff actions are permitted under this Arbitration Agreement. The Arbitrator shall not consolidate claims of different distributors into one proceeding, nor shall the Arbitrator have the power or authority to hear arbitration as a class, collective, representative, or multi-plaintiff action. The Arbitrator may award damages on an individual basis only.

TO THE MAXIMUM EXTENT PERMITTED BY LAW, BOTH PARTIES EXPLICITLY WAIVE ANY **RIGHT TO:** (1) INITIATE OR MAINTAIN ANY COVERED CLAIM ON A CLASS, COLLECTIVE, **REPRESENTATIVE**, OR **MULTI-PLAINTIFF** BASIS EITHER IN COURT OR ARBITRATION: PARTICIPATE AS (2) SERVE OR А REPRESENTATIVE ANY OF SUCH CLASS, COLLECTIVE, OR REPRESENTATIVE ACTION; (3) SERVE OR PARTICIPATE AS A MEMBER OF ANY SUCH CLASS, COLLECTIVE, OR **REPRESENTATIVE ACTION; OR (4) RECOVER** FROM ANY SUCH ANY RELIEF CLASS, COLLECTIVE, REPRESENTATIVE, OR MULTI-PLAINTIFF ACTION.

Any dispute concerning the validity or enforceability of this prohibition against class, collective, representative, or multi-plaintiff action arbitration shall be decided by a court of competent jurisdiction, and no arbitrator shall have any authority to consider or decide any issue concerning the validity or enforceability of such prohibition. Any issues concerning arbitrability of a particular issue or claim under this Arbitration Agreement (except for those concerning the validity or enforceability of the prohibition against class, collective, representative, or multi-plaintiff action arbitration and/or applicability of the FAA) shall be resolved by the arbitrator, not a court.

The Arbitrator shall have the authority to consider and rule on dispositive motions, such as motions to dismiss, or motions for summary judgment, in accordance with the standards and burdens generally applicable to such motions in federal district court, except that the Arbitrator may establish appropriate and less formal procedures for such motions at the Arbitrator's discretion consistent with the expedited nature of arbitration proceedings. The Arbitrator will allow the parties to conduct adequate discovery including, but not limited to, issuing subpoenas to compel the attendance of witnesses at the arbitration hearing; serving written discovery; conducting depositions; and compelling the production of documents during discovery.

Covered Claims covered under this Arbitration Agreement include, but are not limited to: breach of contract, any claims challenging the independent contractor status of DISTRIBUTOR, claims alleging that DISTRIBUTOR was misclassified as an independent other premised contractor. anv claims upon DISTRIBUTOR's alleged status as anything other than an independent contractor, tort claims, discrimination claims, retaliation claims, and claims for alleged unpaid compensation, civil penalties, or statutory penalties under either federal or state law.

This Arbitration Agreement does not cover claims relating to whistleblowers and/or unlawful retaliation arising under the Sarbanes-Oxley Act or disputes involving any ERISA-based benefit plans that provide for arbitration. This Arbitration Agreement also does not preclude either DISTRIBUTOR or COMPANY from seeking provisional remedies such as temporary restraining orders or preliminary injunctions in accordance with applicable law. A party's seeking or obtaining such provisional remedies shall not be considered a waiver of that party's right to arbitration under this Arbitration Agreement.

Nothing in this Arbitration Agreement is intended to affect or limit DISTRIBUTOR's right to file an administrative charge or otherwise seek relief from any administrative or federal or state government agencies (although if DISTRIBUTOR chooses to pursue a claim following the exhaustion of such administrative remedies, that claim would be subject to the provisions of this Arbitration Agreement).

The parties agree that arbitration proceedings are to be treated as confidential, and that the parties will act to protect the confidentiality of the proceedings. The parties agree that neither they nor their counsel will reveal or disclose the substance of the arbitration proceedings, or the result, except as required by subpoena, court order, other legal process, or as otherwise required by law. If disclosure is compelled of one party by subpoena, court order or other legal process, or as otherwise required by law, the party agrees to notify the other party as soon as notice of such process is received and before disclosure takes place. The parties may, however, disclose such information to their legal representatives, accountants or tax advisors as necessary so long as they agree to maintain such information in strict confidence. COMPANY may also disclose such information to individuals in affiliated companies for legal reporting purposes and other legitimate business reasons.

Any request for arbitration must be in writing and provided to the other party and to AAA by certified or registered mail, return receipt requested, within the time period provided for by the statute(s) of limitations applicable to the claim(s) asserted. The request must set forth a statement of the nature of the dispute, including the alleged act or omission at issue; the names of all persons involved in the dispute; the amount in controversy, if any; and the remedy sought.

DISTRIBUTOR acknowledges that this is an important document that affects its legal rights and that the DISTRIBUTOR has been given the opportunity to discuss this Arbitration Agreement with private legal counsel. If any provision of AAA's Rules or of this Arbitration Agreement are determined to be unlawful, invalid, or unenforceable, such provisions shall be enforced to the greatest extent permissible under the law, or, if necessary, severed, and all remaining terms and provisions shall continue in full force and effect. This Arbitration Agreement may be modified or terminated by COMPANY after thirty (30) days written notice to DISTRIBUTOR. Any modifications or terminations shall be prospective only and shall not apply to any claims or disputes that are pending in arbitration or that have been initiated by either party pursuant to the AAA Rules. The parties also agree that nothing herein is intended to, or does, affect or otherwise change the independent contractor relationship between them and that adequate and sufficient consideration has been provided for this Arbitration Agreement, including but not limited to, each party's promise to resolve their claims by arbitration. Finally, this Arbitration Agreement is the complete agreement of the parties on the subject of arbitration of disputes and supersedes any prior or contemporaneous oral or written agreement or understanding on the subject. Any agreement contrary to the foregoing must be in writing signed by the President of COMPANY.

This Arbitration Agreement shall be governed by the FAA and *Connecticut* law to the extent *Connecticut* law is not inconsistent with the FAA.

DISTRIBUTOR acknowledges that it has received and read and specifically agrees to be bound by this Arbitration Agreement. DISTRIBUTOR understands that this Arbitration Agreement requires that disputes that involve matters subject to the Agreement be submitted to arbitration pursuant to the Arbitration Agreement rather than to a judge or jury in court and that such disputes must be brought on an individual basis only.

Date: 4-25-18 (Accepted)

DISTRIBUTOR

COMPANY

<u>s/Tyler Wojnarowski</u> By:Tyler Wojnarowski Its:President Judd E. Price

By:Judd E. Price Its: Market President

OWNER

ATTEST

s/Tyler Wojnarowski

1 JUL

Signature

Tyler Wojnarowski

Print Owner's Name

WITNESS:

La Jac

EXHIBIT N — RECEIPT

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If CK Sales Co., LLC offers you a franchise, it must provide this disclosure document to you 14 calendar-days before you sign a binding agreement with, or make payment to, the franchisor or an affiliate in connection with the proposed franchise sale.

If CK Sales Co., LLC does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal and state law may have occurred and should be reported to the Federal Trade Commission, Washington, DC 20580 and the state agency listed on Exhibit A.

The franchisor is **CK Sales Co., LLC**, located at 11 Adamian Drive, Auburn, Maine 04210. Its telephone number is (207) 783-9161.

Issuance date: April 1, 2018

The franchise seller for this offering is Jake Linthicum, 11 Adamian Drive, Auburn, Maine 04210.

CK Sales Co., LLC authorizes the respective state agencies identified in Exhibit A to receive service of process for it in the particular state.

I received a disclosure document dated *April 1, 2018* that Included the following Exhibits:

Exhibit A — State Regulators; Agents for Service

Exhibit B — Affiliated Companies

Exhibit C — Distributor Agreement.

The Distributor Agreement contains eleven (11) Exhibits:

- (i) Territory
- (ii) Products
- (iii) Authorized Products
- (iv) Bill of Sale
- (v) Owner
- (vi) Personal Guaranty
- (vii) Settlement Statement Authorization
- (viii) Distributor Choice of Insurance Coverages
- (ix) Distributor Choice of Delivery Location
- (x) Distributor Choice of Fee for Service
- (xi) Arbitration Agreement

Exhibit D — Territory Pro Forma

Exhibit E — Financing Documents

- (i) Secured Promissory Note and Personal Guaranty
- (ii) UCC-1

Exhibit F — Form of Purchase Agreement and General Release

Exhibit G — Form of Company Approval of Assignment and General Release

Exhibit H — Form of Assignment/Assumption Agreement and Bill of Sale

Exhibit I — Current Independent Distributors

Exhibit J — Independent Distributor Terminations

Exhibit K — Previous Owner Information

Exhibit L — Financial Statements

Exhibit M — Form of Guarantee

Exhibit N — Receipts

Date:	4-10-2018	Tyler Wojnarowski	s/Tyler Wojnarowski
	(Do not leave blank)	Print Name	Signature

You may return the signed receipt either by signing, dating, and mailing it to CK Sales Co., LLC at 11 Adamian Drive, Auburn, Maine 04210, or by faxing a copy of the signed and dated receipt to Jake Linthicum at (207) 784-4634.

143